DAHLGREN DIVISION NAVAL SURFACE WARFARE CENTER





CSS/TR-02/18

ACOUSTIC SCATTERING FROM LARGE ASPECT RATIO ELASTIC TARGETS

GARY STEVEN SAMMELMANN

LITTORAL WARFARE TECHNOLOGY AND SYSTEMS DEPARTMENT **COASTAL SYSTEMS STATION**

OCTOBER 2002

Approved for public release; distribution is unlimited

20030319 079 **2**9

CSS/TR-02/	18
------------	----

REPORT D	OCUMENTATION	PAGE		Form Approved OBM No. 0704-0188	
Public reporting burden for this collection o maintaining the data needed, and completing and reviewing this burden, to Washington Headquarters Services, Director Budget, Paperwork Reduction Project (0704-0188), Washin	finformation is estimated to average 1 hour per resp the collection of information. Send comments regar ate for Information Operations and Reports, 1215 J gton, DC.	onse, including the time for thing this burden or any o efferson Davis Highway, t	r reviewing instructions, search exis ther aspect of this collection of inforn Suite 1204, Arlington, VA 22202-43	ting data sources, gathering and nation, including suggestions for reducing 12, and to the Office of Management and	
1. AGENCY USE ONLY (Leave blan	(k) 2. REPORT DATE OCTOBER 20	3. REPO	RT TYPE AND DATES	S COVERED	
4. TITLE AND SUBTITLE ACOUSTIC SCATTERING FRO TARGETS	DM LARGE ASPECT RATIO	O ELASTIC	5. FUNDING NU	MBERS	
6. AUTHOR(S)			I		
Gary Steven Samme	elmann				
7. PERFORMING ORGANIZATION NAME(S) AND ADDRESS(ES) ATTN: Code R21 Commanding Officer		8. PERFORMING REPOR	8. PERFORMING ORGANIZATION REPORT NUMBER		
CSS DD NSWC 6703 West Highway 98 Panama City, FL, 32407-7001			CSS/	CSS/TR-02/18	
9. SPONSORING/MONITORING AGENCY NAME(S) AND ADDRESS(ES) 1 Chief of Naval Research 800North Quincy Street Arlington, Virginia 22217-5000 1		10. SPONSORING REPOR	10. SPONSORING/MONITORING AGENCY REPORT NUMBER		
11. SUPPLEMENTARY NOTES			I		
12a. DISTRIBUTION/AVAILABILIT	Y STATEMENT		12b. DISTRIBUT	ION CODE	
Approved for public release; distribution is unlimited			А		
13. ABSTRACT (Maximum 200 words) This paper applies the spher "The Transition Matrix for [Journal of the Acoustic Soc aspect ratio targets. Due to spheroidal wave functions,	o coidal T-matrix approach Acoustic and Elastic Wa ciety of America, 75(1), the limits of double pred these methods are limite	n of Dr. Rog ave Scatterin PP. 35-45] cision arithr d to frequer	ger H Hackman d ng in Prolate Sph to describe the so netic in the comp ncies below kL/2	escribed in his paper eroidal Coordinates", cattering from large putation of the = 30.	
14. SUBJECT TERMS				15. NUMBER OF PAGES	
Personal Computer Shallow Water Acoustic Tool-set (PC SWAT 7.0), Sonar, Minehunting; Mine Countermeasures			106 16. price code		
17. SECURITY CLASSIFICATION 18 OF REPORT	8. SECURITY CLASSIFICATION OF THIS PAGE	19. SECURITY C OF A	LASSIFICATION BSTRACT	20. LIMITATION OF ABSTRACT	
UNCLASSIFIED	UNCLASSIFIED	UNCL	ASSIFIED	SAR	
NSN 7540-01-280-5500				Standard Form 298 (Rev 2-89)	

FOREWORD

This paper applies the spheroidal T-matrix approach of Dr. Roger H Hackman described in his paper "The Transition Matrix for Acoustic and Elastic Wave Scattering in Prolate Spheroidal Coordinates", [Journal of the Acoustic Society of America, 75(1), PP. 35-45] to describe the scattering from large aspect ratio targets. Due to the limits of double precision arithmetic in the computation of the spheroidal wave functions, these methods are limited to frequencies below kL/2 = 30.

This work was funded by Codes 32MIW and 321OE of the Office of Naval Research. The author would like to extend a special thanks to Doug Todoroff, Tom Swean, and Bruce Johnson at the Office of Naval Research for their support.

This report has been reviewed and approved for release by Delbert C. Summey and the Littoral Warfare Technology and Systems Department.

Approved by

١

Delbert C. Summey, Head Littoral Warfare Technology and Systems Department

TABLE OF CONTENTS

.

-

.

.

<u>SECT</u>	SECTION		
I.	INTRODUCTION	1	
II.	SPHEROIDAL COORDINATES		
III.	SPHEROIDAL FUNCTIONS		
IV.	SPHEROIDAL T-MATRIX	9	
V.	SAMPLE CALCULATIONS		
VI.	REFERENCES		
	APPENDICES:		
	A - COMPUTATION OF EIGENVALUES AND EIGENVECTORS	A-1	
	B - EIGENVECTOR ORTHOGONALITY RELATIONS	B-1	
	C - GREEN'S FUNCTION EXPANSION	C-1	
	D - TRANSFORMATION FROM SPHERICAL TO SPHEROIDAL	D-1	
	E - COMPUTATION OF RADIAL FUNCTIONS	E-1	
	F - CONNECTION IN SPHEROIDAL COORDINATES	F-1	
	G - VECTOR BASIS FUNCTIONS	G-1	
	H - PARITY OF BASIS FUNCTIONS	H-1	
	I - COMPUTATION OF T-MATRIX FOR PROLATE SPHEROID	l-1	
	J - COMPUTATION OF T-MATRIX FOR FINITE CYLINDER	J-1	
	DISTRIBUTION	(1)	

.

ILLUSTRATIONS

<u>FIGURE</u>	<u>E</u> <u>P</u>	<u>AGE</u>
1.	FORM FUNCTION AT END-ON INCIDENCE	15
2.	FORM FUNCTION AT 30 DEGREES INCIDENCE	17
3.	FORM FUNCTION AT KL/2 = 5.5	19
4.	BISTATIC FORM FUNCTION AT KL/2 = 5.5 FOR END-ON INCIDENCE	21
5.	TWO-DIMENSIONAL PLOT OF THE FORM FUNCTION AS A FUNCTION OF FREQUENCY VERSUS ASPECT ANGLE	23

I. INTRODUCTION

Prolate spheroidal coordinates are one of eleven coordinate systems in which the scalar Helmholtz Equation in three dimensions is separable. Separability is the reason the scattering from a rigid (sound soft) prolate spheroid may be expressed as a partial wave series of spheroidal wave functions analogous to the partial wave series of a rigid sphere. However, the vector Helmholtz Equation is not separable in the prolate spheroidal coordinate system. This is the main reason for the lack of exact solutions for acoustic scattering from an elastic prolate spheroid. The lack of separability of the vector wave equation in spheroidal coordinates is also the reason there does not exist an expansion of the Green's Dyadic for the vector wave equation in terms of spheroidal functions.

In the early 1980's, Roger Hackman¹⁻³ at Coastal Systems Station used Betti's Identity to derive a T-matrix description of the scattering from large aspect ratio elastic targets based on the spheroidal wave function, rather than the popular spherical T-matrix description. This approach is numerically more stable than the spherical T-matrix approach for large aspect ratio targets. The author had the pleasure of working with Dr. Hackman during his tenure at COASTSYSTA, and worked on the scattering from large aspect ratio targets in a waveguide⁴⁻⁸.

This article focuses on the use of the spheroidal T-matrix description of scattering from elastic targets to describe the low frequency scattering from these targets. The purpose for this work is to build an object oriented computer program for computing the low frequency scattering from large aspect ratio targets. This computer code is to be integrated into the sonar simulation PC Shallow Water Acoustic Tool-set (PC SWAT) to describe a low frequency imaging sonar in the presence of elastic returns from the target.

In addition to the Introduction and Reference sections of this report, the outline of this paper includes the following sections. Section II describes the spheroidal coordinate system and its relationship to linear acoustics. Section III outlines the computation of the spheroidal basis functions. Section IV describes the computation of the spheroidal T-matrix. Section V contains a collection of sample calculations and a comparison of the author's results with those of Hackman. Most of the details of the computations are contained in the appendices. Appendix A describes the computation of the eigenvalues and expansion coefficients for the spheroidal wave equation. Appendix B contains the orthogonality and completeness relationships of the expansion coefficients. Appendix C contains the expansion of the scalar Green's Function and plane wave in terms of spheroidal wave functions. Appendix E describes the computation of the radial spheroidal functions. Appendix F contains a tabulation of the connection components and its first order derivatives used in the computation of the second and third order covariant derivatives of the

scalar basis functions. Appendix G contains a tabulation of the components of the vector basis functions and their covariant derivatives. Appendix H contains a tabulation of the parity of the basis functions. Appendix I describes the computation of the surface integrals and T-matrix in the case of a prolate spheroid. Appendix J describes the computation of the surface integrals and T-matrix, in the case of a finite cylinder with hemi-spherical end caps.

II. SPHEROIDAL COORDINATES

Prolate spheroidal coordinates are a generalization of the familiar spherical coordinates. The transformation from spheroidal to Cartesian coordinates is given by the following equations.

$$x = f\sqrt{(\xi^2 - 1)(1 - \eta^2)} \cos(\varphi)$$

$$y = f\sqrt{(\xi^2 - 1)(1 - \eta^2)} \sin(\varphi)$$

$$z = f\xi\eta$$
(1)

Here, $\xi \ge 1$ is the radial coordinate, $-1 \le \eta \le +1$ is the angular coordinate, $0 \le \varphi \le 2\pi$ is the azimuthal angle coordinate, and f is the semi-focal distance.

Adopt a right-handed coordinate system whose coordinates y^{μ} are given by the spheroidal coordinates (ξ, φ, η) . The metric tensor in this coordinate system is a diagonal metric whose diagonal members are given by the following expressions in terms of the coordinate basis $\{\partial_{\xi}, \partial_{\varphi}, \partial_{\eta}\}$ of the tangent space:

$$g_{\xi\xi} = f^{2}(\xi^{2} - \eta^{2})/(\xi^{2} - 1)$$

$$g_{\eta\eta} = f^{2}(\xi^{2} - \eta^{2})/(1 - \eta^{2})$$

$$g_{\varphi\varphi} = f^{2}(\xi^{2} - 1)(1 - \eta^{2})$$
(2)

Define the ortho-normal triad $\{\hat{e}_1, \hat{e}_2, \hat{e}_3\} = \{\hat{\xi}, \hat{\varphi}, \hat{\eta}\}$ by the following relations:

$$\hat{e}_{i} = e_{i}^{\mu} \partial_{\mu}$$

$$e_{i}^{\mu} = \delta_{i}^{\mu} / h_{\mu}$$

$$h_{\mu} = \sqrt{g_{\mu\mu}}$$
(3)

The covariant derivative of a tensor of type (1,1) in terms of the coordinate basis is of the following form:

$$\nabla_{\mu}T^{\alpha}{}_{\beta} = \partial_{\mu}T^{\alpha}{}_{\beta} + \Gamma^{\alpha}{}_{\sigma\mu}T^{\sigma}{}_{\beta} - \Gamma^{\sigma}{}_{\beta\mu}T^{\alpha}{}_{\sigma} \tag{4}$$

Here, $\Gamma^{\alpha}_{\mu\nu}$ are the components of the torsion-free affine connection defined by the following expression in terms of the metric tensor (An affine transform changes the relationship between the coordinate system a program uses to draw and the coordinate system used to display).

$$\Gamma^{\alpha}_{\ \mu\nu} = \frac{1}{2}g^{\alpha\sigma}(\partial_{\mu}g_{\nu\sigma} + \partial_{\nu}g_{\mu\sigma} - \partial_{\sigma}g_{\mu\nu})$$
⁽⁵⁾

This connection has the property that the covariant derivative of the metric tensor vanishes.

The stress tensor of an elastic solid is defined by the following expression in terms of the metric tensor and the covariant derivative of the displacement vector u.

$$T_{\alpha\beta} = \lambda (\nabla_{\sigma} u^{\sigma}) g_{\alpha\beta} + \mu (\nabla_{\alpha} u_{\beta} + \nabla_{\beta} u_{\alpha})$$
(6)

Here, λ and μ are the Lame constants of an isotropic elastic solid, which are related to the longitudinal and shear wave speeds by the following relations, where ρ is the density of the solid.

$$\lambda = \rho(v_L^2 - 2v_T^2)$$

$$\mu = \rho v_T^2$$
(7)

The equation of motion for an isotropic elastic solid is given by the following expression, where $T(u)_{\alpha\beta}$ is the stress tensor for the displacement vector u_{α} .

$$\rho \partial_{i}^{2} u_{\mu} = \nabla_{\sigma} T^{\sigma}{}_{\mu} \tag{8}$$

A generalization of Green's Theorem gives rise to Betti's Identity, which states that given a pair (u, v) of regular solutions of Equation 8, the following surface integral over the boundary of the enclosed volume (V) vanishes.

$$\oint_{\partial V} ds \{ \vec{u} \bullet \vec{t} (v) - \vec{t} (u) \bullet \vec{v} \} = 0$$
(9)

Here, the terms

$$t(u)_{\mu} = T(u)_{\mu\sigma} n^{\sigma} \tag{10}$$

are the components of the traction of the displacement u.

III. SPHEROIDAL FUNCTIONS

The scalar Helmholtz Equation in spheroidal coordinates is of the following form:

$$(\nabla_{\sigma}\nabla^{\sigma} + k^{2})\psi = \frac{1}{\sqrt{g}}(\partial_{\mu}g^{\mu\nu}\sqrt{g}\partial_{\nu} + k^{2}\sqrt{g})\psi = \frac{1}{f^{2}(\xi^{2} - \eta^{2})}\{\partial_{\xi}(\xi^{2} - 1)\partial_{\xi}\psi + \partial_{\eta}(1 - \eta^{2})\partial_{\eta}\psi + \frac{(\xi^{2} - \eta^{2})}{(\xi^{2} - 1)(1 - \eta^{2})}\partial_{\varphi}^{2}\psi + c^{2}(\xi^{2} - \eta^{2})\psi\} = 0$$
(11)

 $g = Det(g_{\mu\nu})$

Here, c is the dimensionless product of the wavenumber k and the semi-focal distance f. This wave equation is separable into the product of a solution of the radial equation:

$$\partial_{\xi}(\xi^{2}-1)\partial_{\xi}je_{ml} - (\lambda_{ml}(c) - c^{2}\xi^{2} + \frac{m^{2}}{(\xi^{2}-1)})je_{ml} = 0$$
(12)

the angular equation:

$$\partial_{\eta}(1-\eta^2)\partial_{\eta}S_{ml} + (\lambda_{ml}(c) - c^2\eta^2 - \frac{m^2}{(1-\eta^2)})S_{ml} = 0$$
(13)

and the azimuthal equation. The spheroidal coordinates have the rather interesting property that the radial and angular equations are isomorphic under the interchange of the coordinates $\xi \leftrightarrow \eta$. This isomorphism implies a duality between the solutions of the radial and angular functions, that is, the radial functions may be regarded as the analytic continuation of the angular functions to the domain $+1 < \eta < +\infty$. In particular, the radial functions:

$$je_{ml}(c,\xi) = \frac{1}{\kappa_{ml}^{(1)}} S_{ml}^{(1)}(c,\xi)$$

$$ne_{ml}(c,\xi) = \frac{1}{\kappa_{ml}^{(2)}} S_{ml}^{(2)}(c,\xi)$$
(14)

are proportional to the corresponding angular functions. This transformation is an isometry of the metric tensor, that is, the metric tensor is invariant under this transformation. The classic method of solving the angular wave equation is to expand the angular functions of the first kind in terms of associated Legendre polynomials:

$$S_{ml}^{(1)}(c,\eta) = \sum_{k=0,1}^{\infty} d_k^{ml}(c) P_{m+k}^m(\eta)$$
(15)

Here, the sum is over either even or odd indices, depending upon whether the difference (l-m) is even or odd. The series expansion given in Equation 15 is a solution of the angular wave equation provided the expansion coefficients satisfies the following recurrence relations:

$$\alpha_{k}^{m} d_{k+2}^{ml} + (\beta_{k}^{m} - \lambda_{ml}) d_{k}^{ml} + \gamma_{k}^{m} d_{k-2}^{ml} = 0$$

$$\alpha_{k}^{m} = \frac{(2m+k+2)(2m+k+1)c^{2}}{(2m+2k+3)(2m+2k+5)}$$

$$\beta_{k}^{m} = (m+k)(m+k+1)$$

$$+c^{2} \frac{2(m+k)(m+k+1) - 2m^{2} - 1}{(2m+2k-1)(2m+2k+3)}$$

$$m = k(k-1)c^{2}$$
(16)

$$\gamma_k^m = \frac{1}{(2m+2k-3)(2m+2k-1)}$$

The eigenvalue $\lambda_{ml}(c)$ is chosen such that the limit

$$\lim_{k \to \infty} \left(\frac{d_{k+2}^{ml}(c)}{d_{k}^{ml}(c)} \right) = 0$$
(17)

vanishes. The conventional method of solving for the eigenvalue is to solve a transcendental equation defined in terms of a continued fraction of the above eigenvalue and coefficients as described by Flammer in Reference 9. A numerically more robust method of solving for the eigenvalues is to transform the above recurrence relation into the problem of solving for the eigenvalues of an infinite dimensional tri-diagonal matrix. This method is due to Hodge ¹⁰, and is described in References 10 through 12. This is the method adopted by the author since it is readily generalized to the case of complex wavenumber. The coefficients $d_k^{ml}(c)$ are related to the eigenvectors of this infinite dimensional matrix. Reference 13 contains a simple iterative method for solving for the eigenvalues and eigenvectors of arbitrarily large tri-diagonal matrices.

In the case of the radial wave equation, the fact that the radial functions are proportional to the analytic continuation of the angular functions can be used to calculate the radial functions

for radial coordinates near unity $(+1 < \xi < 1.5)$ as described in Reference 11. The computation of the irregular radial functions requires the computation of the angular functions of the second, which is derived by taking the following limit of the series expansion in terms of associated Legendre polynomials of the second kind.

$$S_{ml}^{(2)}(c,\eta) = \lim_{\rho \to 0} \sum_{k=-\infty}^{+\infty} d_{k+\rho}^{ml}(c) Q_{m+k+\rho}^{m}(\eta)$$
(18)

The associated Legendre polynomials of the second kind exhibit a simple pole at integral orders less than -m. At the same time, the coefficients $d_k^{ml}(c)$ exhibit a simple zero, and the product remains finite.

$$\lim_{\rho \to 0} d^{ml}_{-k+\rho} Q^m_{m-k+\rho} = d^{ml}_{k|\rho} P^m_{-m-1+k}, \quad k > +2m$$
⁽¹⁹⁾

This produces a representation of the angular function of the second kind that is a series of the associated Legendre functions of the first and second kind of the following form:

$$S_{ml}^{(2)}(c,\eta) = \sum_{k=-2m+\delta}^{+\infty} d_k^{ml}(c) Q_{m+k}^m(\eta) + \sum_{k=2m+2-\delta}^{+\infty} d_{k|\rho}^{ml}(c) P_{-m-1+k}^m(\eta)$$
(20)

The coefficients $d_{k|\rho}^{ml}(c)$ are defined by the limit:

$$d_{k|\rho}^{ml}(c) = \lim_{\rho \to 0} \frac{d_{-k+\rho}^{ml}(c)}{\rho}$$
(21)

A recurrence relation for these coefficients is given in Reference 11.

A popular representation of the regular radial function in terms of spherical Bessel Functions is of the following form described in References 9 through 12.

$$je_{ml}(c,\xi) = \frac{1}{N_{ml}^{r}} \left(\frac{(\xi^{2}-1)^{1/2}}{\xi}\right)^{m} \sum_{k=0,1}^{\infty} (+i)^{k+m-l} \frac{(2m+k)!}{k!} d_{k}^{ml} j_{m+k}(c\xi)$$
(22)

The above series expansion is absolutely convergent. The corresponding series expansion for the irregular radial function is obtained by replacing the regular spherical Bessel Function with the corresponding irregular spherical Bessel Function. However, in the case of the irregular radial function the series expansion is no longer absolutely convergent, and it is at best an asymptotic expansion.

The author has studied the convergence properties of about a half dozen different representations of the irregular radial functions based on integral representations of the following form:

$$ne_{ml}(c,\xi) = \int d\eta K(\xi,\eta) S_{ml}^{(1)}(c,\eta)$$
(23)

Flammer ⁹ provides a general discussion for the construction of integral representations of the above variety. None of these representations has satisfactory behavior at high frequencies. The high frequency convergence property of the irregular radial functions and the angular functions of the second kind is due to the fact that the expansion coefficients $d_k^{ml}(c)$ represent an alternating series of large terms leading to numerical round off due to the finite precision of the calculations.

IV. SPHEROIDAL T-MATRIX

In the case of the spherical T-matrix description, the expansion of the scalar Green's Function and Green's Dyadic in terms of spherical wave functions is often used in the derivation of the spherical T-matrix description from the scalar and vector Helmholtz Equation. The vector Helmholtz Equation in spheroidal coordinates is non-separable, and there exists no closed form solution for the corresponding Green's Dyadic in terms of spheroidal wave functions. To overcome this difficulty, Dr Hackman made use of Betti's Identity, which is a generalization of Green's Theorem, to generate a system of linear equations for the scattered field in terms of a collection of basis functions of the scalar and vector Helmholtz Equation.

The author introduces the following basis for the regular and outgoing basis functions for the scalar Helmholtz Equation in terms of spheroidal functions:

$$\operatorname{Re} \psi_{\sigma m l} = j e_{m l}(c, \xi) S_{\sigma m l}(c, \eta, \varphi)$$

$$\psi_{\sigma m l} = h e_{m l}(c, \xi) S_{\sigma m l}(c, \eta, \varphi)$$
(24)

The functions $S_{\sigma ml}(c, \eta, \varphi)$ are the following basis functions of the scalar spheroidal harmonics:

$$S_{\sigma m l}(c, \eta, \varphi) = \sqrt{\frac{\varepsilon(m)}{2\pi\Lambda_{m l}(c)}} S_{m l}^{(1)}(c, \eta) \begin{cases} \cos(m\varphi), \sigma = 0\\ \sin(m\varphi), \sigma = 1 \end{cases}$$

$$\varepsilon(m) = \begin{cases} 1, m = 0\\ 2, m \neq 0 \end{cases}$$
(25)

Define the following basis functions of the vector Helmholtz Equation:

$$V_{1,\sigma ml} = \frac{1}{\sqrt{\lambda_{ml}}} \nabla \times (\vec{a} \psi_{\sigma ml}^{T})$$

$$V_{2,\sigma ml} = \frac{1}{c_{T}} \nabla \times V_{1,\sigma ml}$$

$$V_{3,\sigma ml} = \frac{1}{c_{L}} \nabla \psi_{\sigma ml}^{L}$$
(26)

Here, $\lambda_{ml}(c)$ is the eigenvalue of the spheroidal wave equation, and $c_T = k_T f$ and $c_L = k_L f$ are the dimensionless wavenumber for the transverse and longitudinal degrees of freedom. The functions

$$\psi_{\sigma m l}^{T} = h e_{m l}(c_{T},\xi) S_{\sigma m l}(c_{T},\eta,\varphi)$$

$$\psi_{\sigma m l}^{L} = h e_{m l}(c_{L},\xi) S_{\sigma m l}(c_{L},\eta,\varphi)$$
(27)

are the outgoing basis functions for the scalar Helmholtz Equation for the transverse and longitudinal modes. The vector \vec{a} is the conformal killing vector for dilatations whose components are as follows:

$$\vec{a} = (x, y, z), (a_{\xi}, a_{\eta}, a_{\varphi}) = (f^{2}\xi, f^{2}\eta, 0)$$
(28)

This vector satisfies the following condition:

$$\nabla_{\mu}a_{\nu} = g_{\mu\nu} \tag{29}$$

The curl in three dimensions is defined as follows:

$$\nabla \times V = e_{\mu} \varepsilon^{\mu \alpha \beta} \nabla_{\alpha} V_{\beta} \tag{30}$$

in terms of the Levi-Civita Tensor

$$\varepsilon^{\alpha\beta\gamma} = \frac{1}{\sqrt{Det(g_{\mu\nu})}} \delta^{\alpha\beta\gamma}_{123} \tag{31}$$

and the covariant derivative.

Define the vector spheroidal harmonics for the elastic solid as follows:

$$A_{1,\sigma ml} = \frac{1}{\sqrt{\lambda_{ml}}} (\hat{\varphi} \sqrt{1 - \eta^2} \partial_{\eta} - \hat{\eta} \frac{1}{\sqrt{1 - \eta^2}} \partial_{\varphi}) S_{\sigma ml}(c_T, \eta, \varphi)$$

$$A_{2,\sigma ml} = \frac{1}{\sqrt{\lambda_{ml}}} (\hat{\eta} \sqrt{1 - \eta^2} \partial_{\eta} + \hat{\varphi} \frac{1}{\sqrt{1 - \eta^2}} \partial_{\varphi}) S_{\sigma ml}(c_T, \eta, \varphi)$$

$$A_{3,\sigma ml} = \xi S_{\sigma ml}(c_L, \eta, \varphi)$$
(32)

In the limit $f \rightarrow 0$ the vector spheroidal harmonics approach the vector spherical harmonics. They obey the following orthogonality conditions:

$$\begin{split} \oint d\varphi d\eta A_{1,\sigma ml} \bullet A_{1,\sigma' m'l'} &= \delta_{\sigma}^{\sigma'} \delta_{m}^{m'} \Omega_{l,l'}^{m} \\ \oint d\varphi d\eta A_{2,\sigma ml} \bullet A_{2,\sigma' m'l'} &= \delta_{\sigma}^{\sigma'} \delta_{m}^{m'} \Omega_{l,l'}^{m} \\ \oint d\varphi d\eta A_{3,\sigma ml} \bullet A_{3,\sigma' m'l'} &= \delta_{\sigma}^{\sigma'} \delta_{m}^{m'} \delta_{l}^{l'} \\ \oint d\varphi d\eta A_{1,\sigma ml} \bullet A_{3,\sigma' m'l'} &= 0 \\ \oint d\varphi d\eta A_{2,\sigma ml} \bullet A_{3,\sigma' m'l'} &= 0 \\ \oint d\varphi d\eta A_{1,\sigma ml} \bullet A_{2,\sigma' m'l'} &= 0 \end{split}$$

$$(33)$$

$$\Omega_{l,l'}^{m} = \delta_{l}^{l'} - \frac{c_{T}^{2}}{\sqrt{\lambda_{ml} \lambda_{ml'} \Lambda_{ml} \Lambda_{ml'}}} \int_{-1}^{+1} d\eta \eta^{2} S_{ml}(c_{T},\eta) S_{ml'}(c_{T},\eta)$$

The vector basis functions $V_{\tau,\sigma_{ml}}$ have the following asymptotic behavior in the limit $\xi \to +\infty$:

$$V_{1,\sigma ml} \rightarrow he_{ml}(c_T,\xi)A_{1,\sigma ml}$$

$$V_{2,\sigma ml} \rightarrow \frac{1}{c_T}\partial_{\xi}he_{ml}(c_T,\xi)A_{2,\sigma ml}$$

$$V_{3,\sigma ml} \rightarrow \frac{1}{c_L}\partial_{\xi}he_{ml}(c_L,\xi)A_{3,\sigma ml}$$
(34)

The above asymptotic behavior of the vector basis functions coupled with Betti's Identity leads to the following important orthogonality condition for the vector basis functions. Let S be a smooth, closed surface, S_{∞} be the sphere at infinity, V be the volume whose boundaries consists of the union of these two surfaces, and (u, v) be a pair of regular solutions of the vector Helmholtz Equation in this volume. Then Betti's Identity implies the following surface integrals over the closed surface S and the sphere at infinity are equal:

$$\iint_{S} dA\{t(u) \bullet v - u \bullet t(v)\} = \iint_{S_{u}} dA\{t(u) \bullet v - u \bullet t(v)\}$$
(35)

By replacing the functions u and v with the vector basis functions and using the asymptotic expansions of the vector basis functions and Equation 33 to evaluate the surface integral over the sphere at infinity, we obtain the following orthogonality condition for the vector basis functions on the closed surface S:

$$\iint_{S} dA\{t(\operatorname{Re} V_{\tau,\sigma ml}) \bullet \operatorname{Re} V_{\tau,\sigma' m'l'} - \operatorname{Re} V_{\tau,\sigma ml} \bullet t(\operatorname{Re} V_{\tau',\sigma' m'l'})\} = 0$$

$$\iint_{S} dA\{t(V_{\tau,\sigma ml}) \bullet V_{\tau',\sigma' m'l'} - V_{\tau,\sigma ml} \bullet t(V_{\tau',\sigma' m'l'})\} = 0$$

$$\iint_{S} dA\{t(\operatorname{Re} V_{\tau,\sigma ml}) \bullet V_{\tau',\sigma' m'l'} - \operatorname{Re} V_{\tau,\sigma ml} \bullet t(V_{\tau',\sigma' m'l'})\} =$$

$$\begin{cases} -i\mu/k_{\tau}(-i)^{l-l'} \delta_{\sigma}^{\sigma'} \delta_{m}^{m'} \Omega_{l,l'}^{m'}, \tau = \tau' \neq 3 \\ -i(\lambda + 2\mu)/k_{L}(-i)^{l-l'} \delta_{\sigma}^{\sigma'} \delta_{m}^{m'} \delta_{l'}^{l'}, \tau = \tau' = 3 \\ 0, \tau \neq \tau' \end{cases}$$
(36)

The above orthogonality condition is the basis for the spheroidal T-matrix approach, which uses the above orthogonality condition to re-express the boundary conditions of the incident, scattered, and interior fields into a system of linear equations based upon expansions of the incident, scattered and surface fields in terms of these vector basis functions.

Consider the case of acoustic scattering from a solid elastic target immersed in water. The boundary conditions between the exterior and interior fields are given by the following conditions of continuity of normal displacement, continuity of the normal traction, and vanishing of the tangential interior traction.

$$\vec{u}_{+} \bullet \vec{n} = \vec{u}_{-} \bullet \vec{n}$$

$$\vec{t}_{+} \bullet \vec{n} = \vec{t}_{-} \bullet \vec{n}$$

$$\vec{t} \times \vec{n} = 0$$
(37)

The single vector:

$$V_{\sigma m l}^{W} = \frac{1}{c} \nabla h e_{m l}(c, \xi) S_{\sigma m l}(c, \eta, \varphi)$$
(38)

is the vector basis function for the fluid, which may be regarded as the limit $\mu \to 0$ of the vector basis for an elastic medium. The following vector is the corresponding traction vector of this basis:

$$\vec{t}(V_{\sigma ml}^{W}) = \lambda^{W} \vec{n} (\nabla \bullet V_{\sigma ml}^{W}) = -\frac{\lambda^{W} k^{2}}{c} h e_{ml}(c,\xi) S_{\sigma ml}(c,\eta,\varphi) \vec{n}$$
(39)

Make the following expansions of the incident and scattered field in the fluid in terms of this basis:

$$u_{inc} = \sum_{\sigma ml} a_{\sigma ml} \operatorname{Re} V_{\sigma ml}^{W}$$

$$u_{Scatt} = \sum_{\sigma ml} f_{\sigma ml} V_{\sigma ml}^{W}$$
(40)

Applying Betti's Identity to the total field:

$$u_{Total} = u_{inc} + u_{scatt} = \sum_{\sigma ml} a_{\sigma ml} \operatorname{Re} V_{\sigma ml}^{W} + f_{\sigma ml} V_{\sigma ml}^{W}$$
(41)

and the regular and irregular basis functions on the surface S of the scatterer, one arrives at the following linear equations between the expansion coefficients of the incident and scattered fields and the surface fields

$$a_{\sigma m l} = -i \frac{k}{\lambda^{W}} \int_{S} dA\{t_{+} \bullet V_{\sigma m l}^{W} - u_{+} \bullet t(V_{\sigma m l}^{W})\}$$

$$f_{\sigma m l} = +i \frac{k}{\lambda^{W}} \int_{S} dA\{t_{+} \bullet \operatorname{Re} V_{\sigma m l}^{W} - u_{+} \bullet t(\operatorname{Re} V_{\sigma m l}^{W})\}$$
(42)

Using the fact that traction in the fluid is proportional to the normal vector to the surface and the continuity of the normal displacement, one can rewrite these equations in the following form:

$$a_{\sigma ml} = -i\frac{k}{\lambda^{W}} \int_{s} dA\{(t_{+} \bullet n)(n \bullet V_{\sigma ml}^{W}) - (u_{-} \bullet n)n \bullet t(V_{\sigma ml}^{W})\}$$

$$f_{\sigma ml} = +i\frac{k}{\lambda^{W}} \int_{s} dA\{(t_{+} \bullet n)n \bullet \operatorname{Re} V_{\sigma ml}^{W} - (u_{-} \bullet n)n \bullet t(\operatorname{Re} V_{\sigma ml}^{W})\}$$
(43)

Next, one may apply Betti's Identity to the regular vector basis functions of the solid and the surface displacement and traction of the interior fields to arrive at the following equations.

$$\int_{S} dA\{t_{-} \bullet \operatorname{Re} V_{\tau,\sigma ml} - u_{-} \bullet t(\operatorname{Re} V_{\tau,\sigma ml})\} = 0$$
(44)

Use the boundary conditions at the surface to replace this integral with the following:

$$\int_{S} dA\{(t_{+} \bullet n)(n \bullet \operatorname{Re} V_{\tau,\sigma ml}) - u_{-} \bullet t(\operatorname{Re} V_{\tau,\sigma ml})\} = 0$$
(45)

Now make the following expansions in terms of the scalar harmonics in the exterior and the vector harmonics in the interior:

$$t_{+} \bullet n = \sum_{\sigma ml} \alpha_{\sigma ml} S_{\sigma ml}(c, \eta, \varphi)$$

$$u_{-} = \sum_{\tau \sigma ml} \beta_{\tau, \sigma ml} A_{\tau, \sigma ml}$$
(46)

Substituting the above expansions into Equations 43 and 45 one arrives at the following linear system of equations:

$$a = -(Q\beta - M\alpha)$$

$$f = + \operatorname{Re} Q\beta - \operatorname{Re} M\alpha$$

$$0 = P\alpha - R\beta$$

$$Q_{n,\tau'n'} = -i\frac{k}{\lambda^{W}} \int_{S} dA\{(n \bullet t(V_{n}^{W})(n \bullet A_{\tau',n'}))\}$$

$$M_{n,n'} = -i\frac{k}{\lambda^{W}} \int_{S} dA\{(n \bullet V_{n}^{W})S_{n'}\}$$

$$P_{tn,n'} = \int_{S} dA\{(n \bullet \operatorname{Re} V_{\tau,n})S_{n'}\}$$

$$(47)$$

$$R_{tn,\tau'n'} = \int_{S} dA\{t(\operatorname{Re} V_{\tau,n}) \bullet A_{\tau'n'}\}$$

The indices σml have been grouped into the single index n in the above equations. The solution of the above systems of equations is given by the following T-matrix, which relates the incident and scattered fields in the fluid.

$$T = -(\operatorname{Re} QR^{-1}P - \operatorname{Re} M)(QR^{-1}P - M)^{-1}$$
(48)

The above T-matrix is Method B described in Reference 3. This is the method used to perform the calculations in the next section.

V. SAMPLE CALCULATIONS

Figures 1 and 2 depict a comparison of the results of the present author and Dr. Hackman for the scattering from a 4:1 solid aluminum prolate spheroid at end-on incidence (0 degrees) and 30 degrees incidence.



FIGURE 1. FORM FUNCTION AT END-ON INCIDENCE



FIGURE 2. FORM FUNCTION AT 30 DEGREES INCIDENCE

Calculations made by the present author and Dr. Hackman agree within machine precision in the above two cases.

Figure 3 depicts a polar plot of the form function at kL/2=5.5, which corresponds to the frequency of the large flexural wave depicted in Figure 2. The directivity pattern of the flexural resonance at kL/2=5.5 has the characteristic shape of a quadrupole.



FIGURE 3. FORM FUNCTION AT KL/2 = 5.5



Figure 4 depicts a polar plot of the bistatic form function at kL/2=5.5 at end-on incidence. The forward scattering is strongly peaked at the radiation lobes of the flexural wave.

FIGURE 4. BISTATIC FORM FUNCTION AT KL/2 = 5.5 FOR END-ON INCIDENCE

Figure 5 illustrates a two-dimensional rendering of the form function of a 4:1 aluminum spheroid as a function of frequency and aspect angle. The first three flexural resonances of the spheroid are clearly legible.



Aug 16 13:24

FIGURE 5. TWO-DIMENSIONAL PLOT OF THE FORM FUNCTION AS A FUNCTION OF FREQUENCY VERSUS ASPECT ANGLE

VI. REFERENCES

1. Hackman, Roger H., (1984), "The Transition Matrix for Acoustic and Elastic Wave Scattering in Prolate Spheroidal Coordinates", Journal of the Acoustic Society of America, 75(1), PP. 35-45.

2. Hackman, Roger H., (1985), "An Application of the Spheroidal Coordinate Based Transition Matrix: The Acoustic Scattering from High Aspect Ratio Solids", J. Acoust. Soc. Amer., 78(3), pp. 1058-1071.

3. Hackman, Roger H., "Acoustic Scattering from Elastic Solids", in Underwater Scattering and Radiation, Physical Acoustics, Volume 22, edited by A. D. Pierce and R. N. Thurston (Academic Press, San Diego, 1994), pp. 1-194.

4. Lim, Raymond, (1998), "Acoustic Scattering by a Partially Buried Three-Dimensional Elastic Obstacle", J. Acoust. Soc. Amer., 104(2), pp 769-782.

5. Hackman, Roger H., and Gary Steven Sammelmann, (1988)"Multiple Scattering Analysis for a Target in an Oceanic Waveguide", J. Acoust. Soc. Amer., 84(5), pp. 1813-1825.

6. Hackman, Roger H., and Gary Steven Sammelmann, (1986) "Acoustic Scattering in an Inhomogeneous Waveguide: Theory", J. Acoust. Soc. Amer., 80(5), pp. 1447-1458.

7. Hackman, Roger H., and Gary Steven Sammelmann, (1988) "Long Range Scattering in a Deep Oceanic Waveguide: Theory", J. Acoust. Soc. Amer., 8(5), pp. 1776-1793.

8. Hackman, Roger H., and Gary Steven Sammelmann, (1987) "Acoustic Scattering in a Homogeneous Waveguide", J. Acoust. Soc. Amer., 82(1), pp. 324-336.

9. Flammer, C., (1957), Spheroidal Wave Functions, Stanford University Press, Stanford CA.

10. Hodge, D.B., (1970), "Eigenvalues and Eigenfunctions of the Spheroidal Wave Equation", Journal of Mathematics and Physics, V. 11 pp. 2308-2312.

11. Zhang, Shanjie, and Jianming Jin, (1996), *Computation of Special Functions*, John Wiley and Sons Inc., New York.

12. Li, Le-Wei, Xiao-Kang Kang, and Mook-Seng Leong, (2002) Spheroidal Wave Functions in Electromagnetic Theory, John Wiley and Sons Inc., New York.

13. Press, William H., Saul Teukolsky, William Vettering, and Brian Flannery, (1992), Numerical Recipes in Fortran, Cambridge University Press, New York.

APPENDIX A

COMPUTATION OF EIGENVALUES AND EIGENVECTORS

Approved for public release; distribution is unlimited

A-1/A-2

APPENDIX A - COMPUTATION OF EIGENVALUES AND EIGENVECTORS

This appendix describes the computation of the eigenvalues and coefficients d_k^{ml} of the spheroidal wave equation.

The expansion coefficients d_k^{ml} satisfy the following recurrence relation:

$$\alpha_k^m d_{k+2}^{ml} + (\beta_k^m - \lambda_{ml}) d_k^{ml} + \gamma_k^m d_{k-2}^{ml} = 0$$

$$\alpha_k^m = \frac{(2m+k+2)(2m+k+1)c^2}{(2m+2k+3)(2m+2k+5)}$$

$$\beta_k^m = (m+k)(m+k+1) + c^2 \frac{2(m+k)(m+k+1) - 2m^2 - 1}{(2m+2k-1)(2m+2k+3)}$$

$$\gamma_k^m = \frac{k(k-1)c^2}{(2m+2k-3)(2m+2k-1)}$$

Define the ratio

$$N_{k} = -\alpha_{k-2}^{m} \frac{d_{k}^{m}}{d_{k-2}^{m}}$$

These ratios obey the following forward and backward recurrence relations:

$$N_{k+2} = \beta_k^m - \lambda - \frac{\alpha_{k-2}^m \gamma_k^m}{N_k}$$
$$N_k = \frac{\alpha_{k-2}^m \gamma_k^m}{\beta_k^m - \lambda - N_{k+2}}$$

 $N_{2+\delta} = \beta_{\delta}^{m} - \lambda$

$$\delta \equiv (l-m) \operatorname{mod} 2$$

The eigenvalue λ is chosen such that the following limit vanishes:

 $\lim_{k\to\infty} N_k = 0$

The equation for the eigenvalues is obtained by using the forward recurrence relation to compute t he value of the above ratio at k = 2 + l - m. Next, one chooses a sufficiently large value of k = K such that the ratio $N_{K+2} = 0$ vanishes and computes the value of the ratio of k = 2 + l - m from the backward recurrence relation. The eigenvalue is adjusted to make the difference between these two estimates of the ratio at k = 2 + l - m vanish.

Once the eigenvalue has been obtained, the values of the above ratio for $k \le 2+l-m$ are computed from the forward recurrence relation. For indices $k \ge 2+l-m$, the above ratio is calculated from the backward recurrence relation. The rationale for splitting the above calculation into a combination of forward and backward recurrence is to avoid loss of precision by recurring in the direction of increasing magnitude of the ratio. The expansion coefficients d_{k}^{ml} are obtained from the recurrence relation:

$$d_{\delta}^{ml} = 1$$
$$d_{k+2}^{ml} = -\frac{N_{k+2}}{\alpha_k^m} d_k^{ml}$$

The above prescription defines the expansion coefficients d_k^{ml} up to an arbitrary normalization condition. One adopts the normalization condition of References 1 and 2 defined below:

$$S_{ml}^{(1)}(0) = P_l^m(0)$$
, l-m even

 $S_{ml}^{(1)}(0) = P_l^m(0)$, l-m odd

$$\sum_{n=0}^{\infty} (-1)^n \frac{(2n+2m+2\delta)!}{2^{2n+\delta}(n)!(n+m+\delta)!} d_{2n+\delta}^{ml} = (-1)^{(l-m-\delta)/2} \frac{(l+m+\delta)!}{2^{l-m}(\frac{l-m-\delta}{2})!(\frac{l+m+\delta}{2})!}$$

 $\delta \equiv (l-m) \operatorname{mod} 2$

An alternative method of computing the eigenvalues and eigenvectors of the spheroidal wave equations utilizes the fact that the recurrence relation for the expansion coefficients can be represented in the following matrix form as an eigenvalue problem:

$$\begin{pmatrix} \beta_{\delta} & \alpha_{\delta} & 0 & 0 \\ \gamma_{\delta+2} & \beta_{\delta+2} & \alpha_{\delta+2} & 0 \\ 0 & \gamma_{\delta+4} & \beta_{\delta+4} & \alpha_{\delta+4} \\ 0 & 0 & \gamma_{\delta+6} & \beta_{\delta+6} \end{pmatrix} \begin{pmatrix} d_{\delta}^{ml} \\ d_{\delta+2}^{ml} \\ d_{\delta+4}^{ml} \\ d_{\delta+6}^{ml} \end{pmatrix} = \lambda \begin{pmatrix} d_{\delta}^{ml} \\ d_{\delta+2}^{ml} \\ d_{\delta+4}^{ml} \\ d_{\delta+6}^{ml} \end{pmatrix}$$

The matrix can be transformed into a symmetric, tri-diagonal matrix by defining the following expansion coefficients:

 $\Pi_0 \equiv 1, \ n = 0$

$$\Pi_{n} \equiv \sqrt{\frac{\alpha_{\delta+2(n-1)}}{\gamma_{\delta+2n}}} \Pi_{n-1} = \sqrt{\prod_{j=0}^{n-1} \frac{\alpha_{\delta+2j}}{\gamma_{\delta+2(j+1)}}}, \ n \ge 0$$

$$e_0 \equiv d_\delta^{ml}, n \equiv 0$$

$$e_n \equiv \prod_n d_{2n+\delta}^{ml}, \ n > 0$$

These coefficients obey the recurrence relation:

$$a_{n} \equiv \sqrt{\alpha_{\delta+2n}^{m} \gamma_{\delta+2n+2}^{m}}$$
$$b_{n} \equiv \beta_{\delta+2n}^{m}$$
$$a_{0}e_{1} + (b_{0} - \lambda)e_{0} = 0, n = 0$$

$$a_n e_{n+1} + (b_n - \lambda)e_n + a_{n-1}b_{n-1} = 0, n > 0$$

The eigenvalues and the eigenvectors are solutions of the eigenvalue problem for the following symmetric tri-diagonal matrix:

$$\begin{pmatrix} b_0 & a_0 & 0 & 0 \\ a_0 & b_1 & a_1 & 0 \\ 0 & a_1 & b_2 & a_2 \\ 0 & 0 & a_2 & b_3 \end{pmatrix} \begin{pmatrix} e_0 \\ e_1 \\ e_2 \\ e_3 \end{pmatrix} = \lambda \begin{pmatrix} e_0 \\ e_1 \\ e_2 \\ e_3 \end{pmatrix}$$

This method of solving for the eigenvectors is numerically more robust than solving the transcendental equation for the eigenvalues presented previously. In addition, this method is readily generalized to the case the wavenumber (c) is complex. Note: the user is required to solve the above eigenvalue problem for even and odd parity eigenvalues separately.

The value of the expansion coefficients d_k^{ml} for $-2m + \delta \le k \le -2 + \delta$ can be calculated from the backward recurrence relation for the ratio of adjacent coefficients.

 $N_{k} = \frac{\alpha_{k-2}^{m} \gamma_{k}^{m}}{\beta_{k}^{m} - \lambda - N_{k+2}}$ $N_{2+\delta} = \beta_{\delta}^{m} - \lambda$

 $\delta \equiv (l-m) \operatorname{mod} 2$

Alternatively one can use the following forward recurrence relation for the coefficients d_{-k}^{ml} .

$$r_{-k} = -\alpha_{-k}^m / (\beta_{-k}^m - \lambda + \gamma_{-k}^m r_{-k-2})$$

$$r_{-k} = \begin{cases} \frac{d_{-k}^{ml}}{d_{-k+2}^{ml}}, k < 2m + 2 - \delta \\ 0, k \ge 2m + 2 - \delta \end{cases}$$

Since the term $r_{-2m-2+\delta} = 0$ vanishes, the above recurrence relation expresses the ratio as a finite continued fraction. The expansion coefficients d_{-k}^{ml} are given by the following expression:

$$(\prod_{j=0}^{n} r_{-2n-2+\delta}) d_{\delta}^{ml} = d_{-2n-2+\delta}^{ml}, n < m$$

In the case $k \ge -2m-2+\delta$, the coefficients $d_{k+\rho}^{ml}$ have a simple zero in the limit $\rho \to 0$. In this case, one is interested in the coefficients $d_{k|\rho}^{ml}$ for $k \ge -2m-2+\delta$ defined by the relation.

$$d_{k|\rho}^{ml} = \lim_{\rho \to 0} \frac{d_{-k+\rho}^{ml}}{\rho}, k \ge 2m+2-\delta$$

One may use the following forward recurrence relation for the coefficients $d_{k|\rho}^{ml}$:

$$\hat{r}_{-k} = -\hat{\alpha}_{-k}^m / (\beta_{-k}^m - \lambda + \gamma_{-k}^m r_{-k-2}), k \ge 2m + 2 - \delta$$

$$\hat{r}_{-k} \equiv \begin{cases} \frac{d_{2m+2-\delta|\rho}^{ml}}{d_{-2m+\delta}^{ml}}, k = 2m+2-\delta \\ \frac{d_{k|\rho}^{ml}}{d_{k-2|\rho}^{ml}}, k > 2m+2-\delta \end{cases}$$

$$\hat{\alpha}_{-k}^{m} = \begin{cases} \alpha_{-k}^{m}, k \neq 2m + 2 - \delta \\ -\frac{c^{2}}{(2m-1)(2m+1)}, k = 2m + 2 - \delta, \delta = 0 \\ +\frac{c^{2}}{(2m-3)(2m-1)}, k = 2m + 2 - \delta, \delta = 1 \end{cases}$$

Here the coefficients $d_{k|\rho}^{ml}$ are given by the following expression in terms of the ratios \hat{r}_k :

$$(\prod_{j=0}^{n} \hat{r}_{-2(m+n)-2+\delta}) d_{-2m+\delta}^{ml} = d_{+2(m+n)+2-\delta|\rho}^{ml}$$

<u>REFERENCES</u>:

ſ

1. Flammer, C., (1957), Spheroidal Wave Functions, Stanford University Press, Stanford CA.

2. Hodge, D.B., (1970), Eigenvalues and Eigenfunctions of the Spheroidal Wave Equation, Journal of Mathematics and Physics, V. 11 pp. 2308-2312.

APPENDIX B

EIGENVECTOR ORTHOGONALITY RELATIONS

Approved for public release; distribution is unlimited

APPENDIX B - EIGENVECTOR ORTHOGONALITY RELATIONS

This appendix contains the orthogonality relations satisfied by the expansion coefficients d_k^{ml} .

The expansion coefficients d_k^{ml} satisfy the following relationships:

$$\sum_{k=0,1}^{\infty} \frac{2}{(2m+2k+1)} \frac{(2m+k)!}{k!} d_k^{ml} d_k^{ml'} = \Lambda_{ml} \delta_l^{l'}$$

$$\sum_{l=m}^{\infty} d_k^{ml} d_{k'}^{ml} / \Lambda_{ml} = \frac{(2m+2k+1)}{2} \frac{k!}{(2m+k)!} \delta_k^{k'}$$

$$\Lambda_{ml} = \int_{-1}^{+1} d\eta S_{ml}^{(1)}(c,\eta) S_{ml}^{(1)}(c,\eta)$$

The above relationships represent the orthogonality and completeness of the expansion coefficients viewed as eigenvectors of an infinite dimensional tri-diagonal matrix.

APPENDIX C

GREEN'S FUNCTION EXPANSION

Approved for public release; distribution is unlimited

APPENDIX C - GREEN'S FUNCTION EXPANSION

This appendix contains the expansion of the scalar Green's Function and the plane wave in terms of spheroidal wave functions:

$$G(r,r') = \frac{\exp[+ik|r-r'|]}{4\pi|r-r'|} = ik\sum_{\sigma ml} he_{ml}(c,\xi_{>}) je_{ml}(c,\xi_{<})S_{\sigma ml}(\eta,\varphi)S_{\sigma ml}(\eta',\varphi')$$

 $\exp[+i\vec{k} \cdot \vec{r}] = \exp[+ikr\cos(\gamma)] = 4\pi \sum_{\sigma ml} (+i)^l j e_{ml}(c,\xi) S_{\sigma ml}(\eta,\varphi) S_{\sigma ml}(\eta',\varphi')$

 $\cos(\gamma) = \cos(\vartheta)\cos(\vartheta') + \sin(\vartheta)\sin(\vartheta')\cos(\varphi - \varphi')$

APPENDIX D

TRANSFORMATION FROM SPHERICAL TO SPHEROIDAL

Approved for public release; distribution is unlimited

x.

D-1/D-2
APPENDIX D - TRANSFORMATION FROM SPHERICAL TO SPHEROIDAL

This appendix contains the transformation between spherical and spheroidal wave functions.

The transformation between spherical and spheroidal wave functions is given by the following transformation, where the matrix $B_{\sigma ml,\sigma'm'l'}$ is a unitary matrix:

$$he_{ml}S_{\sigma ml} = \sum_{\sigma'm'l'} B_{\sigma ml,\sigma'm'l'} h_{l'}Y_{\sigma'm'l'}$$

$$h_l Y_{\sigma m l} = \sum_{\sigma' m' l'} B^{-1}{}_{\sigma m l, \sigma' m' l'} h e_{m' l'} S_{\sigma' m' l}$$

$$B_{\sigma m l, \sigma' m' l'} = \delta_{\sigma}^{\sigma'} \delta_{m}^{m'} \frac{(+i)^{l'-l}}{\sqrt{\Lambda_{m l}}} \sqrt{\frac{2}{(2l'+1)} \frac{(l'+m)!}{(l'-m)!}} d_{l'-m}^{m l}$$

$$B^{-1}_{\sigma_{ml},\sigma'm'l'} = \delta_{\sigma}^{\sigma'}\delta_{m}^{m'}\frac{(+i)^{l'-l}}{\sqrt{\Lambda_{ml'}}}\sqrt{\frac{2}{(2l+1)}\frac{(l+m)!}{(l-m)!}}d_{l-m}^{ml'}$$

$$B^{-1}_{\sigma ml,\sigma'm'l'} = B^{\dagger}_{\sigma ml,\sigma'm'l'}$$

APPENDIX E

COMPUTATION OF RADIAL FUNCTIONS

Approved for public release; distribution is unlimited

Ľ

E-1/E-2

APPENDIX E - COMPUTATION OF RADIAL FUNCTIONS

This appendix describes various representations of the radial functions. These representations are based on integral representations of the following form described by Flammer in Reference 1.

$$R_{ml}(c,\xi) = \int_{a}^{b} d\eta K(\xi,\eta) S_{ml}^{(1)}(c,\eta)$$

The following Theorem and Lemma are from Reference 1:

THEOREM:

Let L_{ξ} and L_{η} be the following differential operators:

$$L_{\xi} = \partial_{\xi} (\xi^{2} - 1) \partial_{\xi} + c^{2} \xi^{2} - \frac{m^{2}}{(\xi^{2} - 1)}$$
$$L_{\eta} = \partial_{\eta} (1 - \eta^{2}) \partial_{\eta} - c^{2} \eta^{2} - \frac{m^{2}}{(1 - \eta^{2})}$$

Let $K(\xi,\eta)$ be a solution of the following partial differential equation in the complex domain D:

$$(L_{\xi}-L_{\eta})K(\xi,\eta)=0$$

Let [a,b] be the contour in the complex domain D such that the following integral vanishes:

$$\int_{a}^{b} d\eta \{ (L_{\eta}K(\xi,\eta)) S_{ml}^{(1)}(c,\eta) - K(\xi,\eta) L_{\eta} S_{ml}^{(1)}(c,\eta) \} =$$

$$(1-\eta^{2}) \{ (\partial_{\eta}K(\xi,\eta)) S_{ml}^{(1)}(c,\eta) - K(\xi,\eta) \partial_{\eta} S_{ml}^{(1)}(c,\eta) \}_{a}^{b} = 0$$

Then the following contour integral is a solution of the radial equation in the complex domain D:

$$R_{ml}(c,\xi) = \int_{a}^{b} d\eta K(\xi,\eta) S_{ml}^{(1)}(c,\eta)$$
$$(L_{\xi} + \lambda_{ml}) R_{ml} = 0$$

LEMMA:

Suppose:

$$\psi(\xi,\eta,\varphi) = K(\xi,\eta) \exp[+im\varphi]$$

is a solution of the Helmholtz Wave Equation.

$$(\nabla^2 + k^2)\psi = 0$$

Then the function $K(\xi,\eta)$ is a solution of the partial differential equation:

$$(L_{\xi}-L_{\eta})K(\xi,\eta)=0$$

Some of the examples of kernels satisfying the above partial differential equation listed in Reference 1 are listed below:

$$\begin{aligned} (\xi^{2}-1)^{m/2}(1-\eta^{2})^{m/2} \exp[+ic\xi\eta] \\ J_{m}(c\sqrt{(\xi^{2}-1)(1-\eta^{2})}) \\ \xi\eta J_{m}(c\sqrt{(\xi^{2}-1)(1-\eta^{2})}) \\ J_{m}(c\sqrt{(\xi^{2}-1)(1-\eta^{2})} \sin(\alpha)) \exp[\pm ic\xi\eta\cos(\alpha)] \\ j_{m}(c(\xi\pm\eta))(\xi^{2}-1)^{m/2}(1-\eta^{2})^{m/2}(c(\xi\pm\eta))^{-m} \\ h^{(1)}{}_{m}(c(\xi\pm\eta))(\xi^{2}-1)^{m/2}(1-\eta^{2})^{m/2}(c(\xi\pm\eta))^{-m} \end{aligned}$$

The classic expansion ¹⁻² of the radial functions in terms of spherical Bessel Functions follows:

$$je_{ml}(c,\xi) = \frac{1}{\Lambda_{ml}^{(r)}} \frac{(\xi^2 - 1)^{m/2}}{\xi^m} \sum_{n=0}^{\infty} (+i)^{2n+\delta+m-l} \frac{(2m+2n+\delta)!}{(2n+\delta)!} d_{2n+\delta}^{ml} j_{2n+m+\delta}(c\xi)$$

$$ne_{ml}(c,\xi) = \frac{1}{\Lambda_{ml}^{(r)}} \frac{(\xi^2 - 1)^{m/2}}{\xi^m} \sum_{n=0}^{\infty} (+i)^{2n+\delta+m-l} \frac{(2m+2n+\delta)!}{(2n+\delta)!} d_{2n+\delta}^{ml} n_{2n+m+\delta}(c\xi)$$

$$\delta = (l-m) \mod 2$$

$$\Lambda_{ml}^{(r)} = \sum_{n=0}^{\infty} \frac{(2n+2m+\delta)!}{(2n+\delta)!} d_{2n+\delta}^{ml}$$

The expansion for the regular function je_{ml} is absolutely convergent, whereas the expansion for the irregular function ne_{ml} is an asymptotic expansion unsuitable for radial coordinates near unity.

The expansions of the radial functions in terms of the analytic continuation of the angular functions are given by the following expressions:

$$je_{ml}(c,\xi) = \frac{1}{\kappa_{ml}^{(1)}} S_{ml}^{(1)}(c,\xi)$$

$$ne_{ml}(c,\xi) = \frac{1}{\kappa_{ml}^{(2)}} S_{ml}^{(2)}(c,\xi)$$

$$\kappa_{ml}^{(1)} = + \frac{(2m+2\delta+1)(l+m+\delta)!}{2^{l+m}c^{m+\delta}d_{\delta}^{ml}m!(\frac{l-m-\delta}{2})!(\frac{l+m+\delta}{2})!} \sum_{n=0}^{\infty} \frac{(2m+2n+\delta)!}{(2n+\delta)!} d_{2n+\delta}^{ml}$$

$$\kappa_{ml}^{(2)} = + \frac{2^{l-m}(2m)!(\frac{l-m}{2})!(\frac{l+m}{2})!d_{-2m}^{ml}}{(2m-1)m!(l+m)!c^{m-1}} \sum_{n=0}^{\infty} \frac{(2m+2n)!}{(2n)!} d_{2n}^{ml}, \text{ even (l-m)}$$

$$\kappa_{ml}^{(2)} = -\frac{2^{l-m}(2m)!(\frac{l-m-1}{2})!(\frac{l+m+1}{2})!d_{-2m+1}^{ml}}{(2m-1)(2m-3)m!(l+m+1)!c^{m-2}} \sum_{n=0}^{\infty} \frac{(2m+2n+\delta)!}{(2n+\delta)!} d_{2n+\delta}^{ml}, \text{ odd (l-m)}$$

The above representation of the irregular radial function ne_{ml} in terms of associated Legendre polynomials is useful for calculating the radial function for arguments in the range $+1 < \xi \le 1.5$. In the above expressions, one uses the analytic continuation of the associated Legendre polynomial to the complex plane:

$$P_{l}^{m}(z) = (z^{2} - 1)^{m/2} (\frac{d}{dz})^{m} P_{l}(z)$$
$$Q_{l}^{m}(z) = (z^{2} - 1)^{m/2} (\frac{d}{dz})^{m} Q_{l}(z)$$

The following expansion of the regular radial function in terms of spherical Bessel Functions is sometimes useful for arguments near unity:

$$je_{ml} = \frac{1}{S_{ml}^{(1)}(0)} \sum_{k} (+i)^{k+m-l} d_k^{ml} P_{m+k}^m(0) j_{m+k} (c\sqrt{\xi^2 - 1}) \quad (\text{even l-m})$$

$$je_{ml} = \frac{1}{S_{ml}^{(1)}(0)} \frac{\xi}{\sqrt{\xi^2 - 1}} \sum_{k} (+i)^{k+m-l} d_k^{ml} P_{m+k}^{(m)}(0) j_{m+k} (c\sqrt{\xi^2 - 1}) \text{ (odd l-m)}$$

There is an error in Reference 1 in the case l-m is odd in the above expansion.

The radial functions have the following expansions in terms of the product of spherical Bessel Functions:

 $\xi = \cosh(\mu/2)$

Even l-m

1

$$he_{ml} = (\xi^2 - 1)^{m/2} \sum_{n=0}^{\infty} (2n + 2m + 1) D_n^{ml} h_{m+n}^{(1)} (ce^{+\mu/2}) j_{m+n} (ce^{-\mu/2})$$
$$D_n^{ml} = \frac{(+i)^{l-m} 2^m (n+2m)!}{d_0^{ml} c^m m! n!} \sum_{k=0}^n d_{2k}^{ml} \frac{(-n)_k (n+2m+1)_k (m+1/2)_k}{k!} \frac{\Gamma(m+3/2)}{\Gamma(m+3/2+2k)}$$

$$\times_{3}F_{2}(-n+k, n+k+2m+1, k+1/2; m+k+1, m+2k+3/2; +1)$$

$$=\frac{(+i)^{l-m}(2m+1)2^{m-1}(n+2m)!}{d_0^{ml}c^mm!n!}\sum_{k=0}^n d_{2k}^{ml}\frac{(-n)_k(n+2m+1)_k(m+1/2)_k}{k!}\frac{\Gamma(m+n-k+1/2)}{\Gamma(m+n+k+3/2)}$$

Odd l-m

$$he_{ml} = \xi(\xi^2 - 1)^{m/2} \sum_{n=0}^{\infty} (2n + 2m + 3)h_{m+n+1}^{(1)}(ce^{+\mu/2}) j_{m+n+1}(ce^{-\mu/2}) D_n^{ml}$$

$$D_n^{ml} = \frac{(+i)^{l-m-1}(2m+3)2^m(2m+n+2)!}{d_1^{ml}c^{m+1}m!n!} \sum_{k=0}^n d_{2k+1}^{ml} \frac{(-n)_k(m+3/2)_k(n+2m+3)_k}{k!} \frac{\Gamma(m+3/2)}{\Gamma(m+5/2+2k)}$$

 $\times_{3}F_{2}(-n+k,k+3/2,n+2m+k+3;m+k+2,m+2k+5/2:+1)$

The Pochammer and generalized Hypergeometric function are defined below:

$$(x)_{k} = \Gamma(x+k) / \Gamma(x) = \prod_{n=0}^{k-1} (x+n)$$

$${}_{3}F_{2}(a_{1}, a_{2}, a_{3}; b_{1}, b_{2} : z) = \sum_{n=0}^{\infty} \frac{(a_{1})_{n} (a_{2})_{n} (a_{3})_{n}}{n! (b_{1})_{n} (b_{2})_{n}} z^{n}$$

REFERENCES:

1. Flammer, C., (1957), Spheroidal Wave Functions, Stanford University Press, Stanford CA.

2. Hodge, D.B., (1970), Eigenvalues and Eigenfunctions of the Spheroidal Wave Equation, Journal of Mathematics and Physics, V. 11 pp. 2308-2312

APPENDIX F

CONNECTION IN SPHEROIDAL COORDINATES

Approved for public release; distribution is unlimited

¥

APPENDIX F - CONNECTION IN SPHEROIDAL COORDINATES

This appendix contains a tabulation of the non-zero terms of the linear connection, its derivatives, and the function $P^{\mu}_{\alpha\beta\gamma} = \partial_{\alpha}\Gamma^{\mu}_{\beta\gamma} + \Gamma^{\mu}_{\sigma\alpha}\Gamma^{\sigma}_{\beta\gamma}$ in prolate spheroidal coordinates. These functions are used to compute the second and third order covariant derivatives of the scalar basis functions used in the construction of the vector basis and the stress tensor:

$$\begin{split} \Gamma_{\xi\xi}^{\xi} &= -\frac{\xi(1-\eta^2)}{(\xi^2-1)(\xi^2-\eta^2)} \\ \Gamma_{\eta\eta}^{\eta} &= +\frac{\eta(\xi^2-1)}{(1-\eta^2)(\xi^2-\eta^2)} \\ \Gamma_{\xi\eta}^{\xi} &= \Gamma_{\eta\xi}^{\xi} = -\frac{\eta}{(\xi^2-\eta^2)} \\ \Gamma_{\eta\xi}^{\xi} &= \Gamma_{\eta\xi}^{\eta} = +\frac{\xi}{(\xi^2-\eta^2)} \\ \Gamma_{\eta\eta}^{\xi} &= \Gamma_{\xi\eta}^{\eta} = +\frac{\xi}{(\xi^2-\eta^2)} \\ \Gamma_{\xi\xi}^{\xi} &= +\frac{\eta(1-\eta^2)}{(1-\eta^2)(\xi^2-\eta^2)} \\ \Gamma_{\varphi\xi}^{\varphi} &= \Gamma_{\eta\varphi}^{\varphi} = -\frac{\eta}{(1-\eta^2)} \\ \Gamma_{\varphi\xi\xi}^{\varphi} &= \Gamma_{\xi\varphi}^{\varphi} = +\frac{\xi}{(\xi^2-1)} \\ \Gamma_{\varphi\varphi\xi}^{\xi} &= \Gamma_{\xi\varphi}^{\varphi} = +\frac{\xi}{(\xi^2-\eta^2)} \\ \Gamma_{\varphi\varphi\varphi}^{\eta} &= -\frac{\eta(1-\eta^2)(\xi^2-\eta^2)}{(\xi^2-\eta^2)} \\ \Gamma_{\varphi\varphi\varphi}^{\eta} &= -\frac{\eta(1-\eta^2)(\xi^2-1)}{(\xi^2-\eta^2)} \\ \partial_{\xi}\Gamma_{\xi\xi}^{\xi} &= -\frac{(1-\eta^2)}{(\xi^2-\eta^2)^2} \{1 - \frac{2\xi^2}{(\xi^2-\eta^2)^2}\} \\ \partial_{\eta}\Gamma_{\xi\xi}^{\xi} &= \frac{2\xi\eta}{(\xi^2-\eta^2)^2} \end{split}$$

$$\begin{split} \partial_{\xi}\Gamma_{\xi\eta}^{\xi} &= \partial_{\xi}\Gamma_{\eta\xi}^{\xi} = + \frac{2\xi\eta}{(\xi^{2} - \eta^{2})^{2}} \\ \partial_{\eta}\Gamma_{\xi\eta}^{\xi} &= \partial_{\eta}\Gamma_{\eta\xi}^{\xi} = -\frac{(\xi^{2} + \eta^{2})}{(\xi^{2} - \eta^{2})^{2}} \\ \partial_{\xi}\Gamma_{\eta\eta}^{\xi} &= + \frac{1}{(1 - \eta^{2})(\xi^{2} - \eta^{2})} \{1 - 3\xi^{2} + \frac{2\xi^{2}(\xi^{2} - 1)}{(\xi^{2} - \eta^{2})}\} \\ \partial_{\eta}\Gamma_{\eta\eta}^{\xi} &= -\frac{2\xi\eta(\xi^{2} - 1)}{(1 - \eta^{2})(\xi^{2} - \eta^{2})} \{\frac{1}{(1 - \eta^{2})} + \frac{1}{(\xi^{2} - \eta^{2})}\} \\ \partial_{\xi}\Gamma_{\varphi\varphi}^{\xi} &= + \frac{(1 - \eta^{2})}{(\xi^{2} - \eta^{2})} \{1 - 3\xi^{2} + \frac{2\xi^{2}(\xi^{2} - 1)}{(\xi^{2} - \eta^{2})}\} \\ \partial_{\eta}\Gamma_{\varphi\varphi}^{\xi} &= \partial_{\xi}\Gamma_{\xi\varphi}^{\varphi} = -\frac{(\xi^{2} + 1)}{(\xi^{2} - \eta^{2})^{2}} \\ \partial_{\eta}\Gamma_{\varphi\xi}^{\varphi} &= \partial_{\xi}\Gamma_{\xi\varphi}^{\varphi} = 0 \\ \partial_{\xi}\Gamma_{\varphi\eta}^{\varphi} &= \partial_{\xi}\Gamma_{\eta\varphi}^{\varphi} = 0 \\ \partial_{\xi}\Gamma_{\eta\eta}^{\varphi} &= \partial_{\eta}\Gamma_{\eta\varphi}^{\varphi} = -\frac{(1 + \eta^{2})}{(1 - \eta^{2})^{2}} \\ \partial_{\xi}\Gamma_{\eta\eta}^{\eta} &= +\frac{2\xi\eta}{(\xi^{2} - \eta^{2})^{2}} \\ \partial_{\eta}\Gamma_{\eta\eta}^{\eta} &= +\frac{(\xi^{2} - 1)}{(1 - \eta^{2})(\xi^{2} - \eta^{2})} \{1 + \frac{2\eta^{2}}{(1 - \eta^{2})} + \frac{2\eta^{2}}{(\xi^{2} - \eta^{2})}\} \\ \partial_{\xi}\Gamma_{\eta\xi}^{\eta} &= \partial_{\xi}\Gamma_{\xi\eta}^{\eta} = -\frac{(\xi^{2} + \eta^{2})}{(\xi^{2} - \eta^{2})^{2}} \\ \partial_{\eta}\Gamma_{\eta\xi}^{\eta} &= \partial_{\xi}\Gamma_{\xi\eta}^{\eta} = -\frac{(\xi^{2} + \eta^{2})}{(\xi^{2} - \eta^{2})^{2}} \\ \partial_{\eta}\Gamma_{\eta\xi}^{\eta} &= \partial_{\xi}\Gamma_{\xi\eta}^{\eta} = -\frac{(\xi^{2} + \eta^{2})}{(\xi^{2} - \eta^{2})^{2}} \end{split}$$

۰ť

$$\begin{split} \partial_{\xi} \Gamma^{\eta}_{\xi\xi} &= -\frac{2\xi\eta(1-\eta^{2})}{(\xi^{2}-1)(\xi^{2}-\eta^{2})} \{\frac{1}{(\xi^{2}-1)} + \frac{1}{(\xi^{2}-\eta^{2})} \} \\ \partial_{\eta} \Gamma^{\eta}_{\xi\xi} &= +\frac{1}{(\xi^{2}-1)(\xi^{2}-\eta^{2})} \{1-3\eta^{2} + \frac{2\eta^{2}(1-\eta^{2})}{(\xi^{2}-\eta^{2})} \} \\ \partial_{\xi} \Gamma^{\eta}_{\varphi\varphi} &= +\frac{2\xi\eta(1-\eta^{2})^{2}}{(\xi^{2}-\eta^{2})^{2}} \\ \partial_{\eta} \Gamma^{\eta}_{\varphi\varphi\varphi} &= +\frac{(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \{1-3\eta^{2} + \frac{2\eta^{2}(1-\eta^{2})}{(\xi^{2}-\eta^{2})} \} \\ P^{\xi}_{\xi\xi\xi} &= +\frac{3\xi^{2}(1-\eta^{2})}{(\xi^{2}-1)^{2}(\xi^{2}-\eta^{2})} \\ P^{\eta}_{\xi\xi\xi} &= -\frac{\xi\eta}{(\xi^{2}-1)(\xi^{2}-\eta^{2})} \\ P^{\eta}_{\eta\xi\xi} &= -\frac{\eta^{2}}{(\xi^{2}-1)(\xi^{2}-\eta^{2})} \\ P^{\xi}_{\xi\xi\eta} &= P^{\xi}_{\xi\eta\xi} &= +\frac{\xi\eta}{(\xi^{2}-1)(\xi^{2}-\eta^{2})} \\ P^{\xi}_{\xi\xi\eta} &= P^{\xi}_{\xi\eta\xi} &= -\frac{\xi^{2}}{(1-\eta^{2})(\xi^{2}-\eta^{2})} \\ P^{\xi}_{\eta\xi\eta} &= P^{\eta}_{\eta\eta\xi} &= -\frac{\xi^{2}}{(\xi^{2}-1)(\xi^{2}-\eta^{2})} \\ P^{\eta}_{\xi\eta\eta} &= -\frac{\xi^{2}}{(\xi^{2}-1)(\xi^{2}-\eta^{2})} \\ P^{\eta}_{\xi\eta\eta} &= -\frac{\xi^{2}}{(1-\eta^{2})(\xi^{2}-\eta^{2})} \\ P^{\xi}_{\xi\eta\eta} &= -\frac{\xi^{2}}{(1-\eta^{2})(\xi^{2}-\eta^{2})} \\ P^{\eta}_{\xi\eta\eta} &= -\frac{\xi^{2}}{(1-\eta^{2})(\xi^{2}-\eta^{2})} \\ P^{\xi}_{\xi\eta\eta} &= -\frac{\xi^{2}}{(1-\eta^{2})(\xi^{2}-\eta^{2})} \\ P^{\eta}_{\xi\eta\eta} &= -\frac{\xi^{2}}{(1-\eta^{2})(\xi^{2}-\eta^{2})} \\ P^{\eta}_{\xi\eta\eta} &= -\frac{\xi^{2}}{(1-\eta^{2})(\xi^{2}-\eta^{2})} \\ P^{\eta}_{\xi\eta\eta} &= -\frac{3\xi\eta(\xi^{2}-1)}{(1-\eta^{2})^{2}(\xi^{2}-\eta^{2})} \\ \end{array}$$

¥

$$P^{\varphi}_{\varphi\xi\xi} = -\frac{1}{(\xi^2 - 1)^2}$$

$$P^{\varphi}_{\varphi\xi\eta} = P^{\varphi}_{\varphi\eta\xi} = -\frac{\xi\eta}{(1 - \eta^2)(\xi^2 - 1)}$$

$$P^{\varphi}_{\varphi\eta\eta} = -\frac{1}{(1 - \eta^2)^2}$$

$$P^{\varphi}_{\varphi\varphi\varphi} = -1$$

.

APPENDIX G

VECTOR BASIS FUNCTIONS

Approved for public release; distribution is unlimited

*

G-1/G-2

APPENDIX G - VECTOR BASIS FUNCTIONS

This appendix contains the components of the vector basis functions and their covariant derivatives for an elastic medium in spheroidal coordinates. One may use the index n to signify the group of indices σml . Here, ψ_n^T and ψ_n^L are the scalar basis functions for the transverse and longitudinal modes. The vector basis functions are defined by the following relationships.

$$\begin{aligned} a_{\xi} &= f^{2}\xi \\ a_{\eta} &= f^{2}\eta \\ a_{\varphi} &= 0 \end{aligned}$$

$$g &= Det(g_{\mu\nu}) = f^{6}(\xi^{2} - \eta^{2})^{2} \\ \sqrt{\lambda}(V_{n}^{1})_{\mu} &= -\frac{g_{\mu\nu}\delta_{\xi\varphi\eta}^{\nu\alpha\beta}a_{\alpha}\nabla_{\beta}\psi_{n}^{T}}{\sqrt{g}} \\ k_{\tau}\sqrt{\lambda}(V_{n}^{2})_{\mu} &= k_{\tau}^{2}a_{\mu}\psi_{n}^{T} + 2\nabla_{\mu}\psi_{n}^{T} + a^{\sigma}\nabla_{\sigma}\nabla_{\mu}\psi_{n}^{T} \\ k_{L}(V_{n}^{3})_{\mu} &= \nabla_{\mu}\psi_{n}^{L} \\ \sqrt{\lambda_{n}}\nabla_{\mu}(V_{n}^{1})_{\nu} &= +\frac{g_{\mu\sigma}g_{\nu\alpha}\delta_{\xi\varphi\eta}^{\alpha\alpha\beta}\nabla_{\beta}\psi_{n}^{T} - g_{\nu\sigma}\delta_{\xi\varphi\eta}^{\alpha\alpha\beta}a_{\alpha}\nabla_{\beta}\nabla_{\mu}\psi_{n}^{T}}{\sqrt{g}} \\ k_{\tau}\sqrt{\lambda_{n}}\nabla_{\mu}(V_{n}^{1})_{\nu} &= k_{\tau}^{2}g_{\mu\nu}\psi_{n}^{T} + k_{\tau}^{2}a_{\nu}\nabla_{\mu}\psi_{n}^{T} + 3\nabla_{\mu}\nabla_{\nu}\psi_{n}^{T} + a^{\sigma}\nabla_{\sigma}\nabla_{\mu}\nabla_{\nu}\psi_{n}^{T} \\ \nabla_{\mu}(V_{n}^{3})_{\nu} &= \frac{1}{k_{L}}\nabla_{\mu}\nabla_{\nu}\psi_{n}^{L} \end{aligned}$$

First, one tabulates the components of the second and third order covariant derivatives of the scalar basis functions. These functions are used to construct the vector basis and its covariant derivative. Since the Riemann Curvature Tensor vanishes, the covariant derivatives commute in a coordinate basis. Thus, the higher order covariant derivatives of the scalar basis functions are symmetric tensors. The radial, angular, and azimuthal wave equations have been used to eliminate second and higher order derivatives with respect to a given variable:

$$\begin{split} \nabla_{\xi} \nabla_{\xi} \psi &= -\frac{\eta(1-\eta^{2})}{(\xi^{2}-1)(\xi^{2}-\eta^{2})} \partial_{\eta} \psi - \frac{\xi}{(\xi^{2}-1)} [2 - \frac{(1-\eta^{2})}{(\xi^{2}-\eta^{2})}] \partial_{\xi} \psi + \frac{1}{(\xi^{2}-1)} [\lambda - c^{2}\xi^{2} + \frac{m^{2}}{(\xi^{2}-1)}] \psi \\ \nabla_{\eta} \nabla_{\eta} \psi &= + \frac{\xi(\xi^{2}-1)}{(1-\eta^{2})(\xi^{2}-\eta^{2})} \partial_{\xi} \psi + \frac{\eta}{(1-\eta^{2})} [2 - \frac{(\xi^{2}-1)}{(\xi^{2}-\eta^{2})}] \partial_{\eta} \psi - \frac{1}{(1-\eta^{2})} [\lambda - c^{2}\eta^{2} - \frac{m^{2}}{(1-\eta^{2})}] \psi \\ \nabla_{\xi} \nabla_{\eta} \psi &= \partial_{\xi} \partial_{\eta} \psi + \frac{1}{(\xi^{2}-\eta^{2})} [\eta \partial_{\xi} \psi - \xi \partial_{\eta} \psi] \\ \nabla_{\xi} \nabla_{\varphi} \psi &= \partial_{\xi} \partial_{\varphi} \psi - \frac{\xi}{(\xi^{2}-1)} \partial_{\varphi} \psi \\ \nabla_{\eta} \nabla_{\varphi} \psi &= \partial_{\eta} \partial_{\varphi} \psi + \frac{\eta}{(1-\eta^{2})} \partial_{\varphi} \psi \\ \nabla_{\eta} \nabla_{\varphi} \psi &= -m^{2} \psi + \frac{(1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} [\xi \partial_{\xi} \psi - \eta \partial_{\eta} \psi] \\ \nabla_{\xi} \nabla_{\xi} \nabla_{\xi} \nabla_{\xi} \psi &= -\frac{3\eta(1-\eta^{2})}{(\xi^{2}-1)(\xi^{2}-\eta^{2})} \partial_{\xi} \partial_{\eta} \psi \\ + \frac{2\xi\eta(1-\eta^{2})}{(\xi^{2}-1)(\xi^{2}-\eta^{2})} [\frac{2}{(\xi^{2}-\eta^{2})} + \frac{1}{(\xi^{2}-1)} - \frac{(1-\eta^{2})}{(\xi^{2}-1)(\xi^{2}-\eta^{2})}] \partial_{\eta} \psi \\ + \frac{3(1-\eta^{2})}{(\xi^{2}-\eta^{2})^{2}} [\xi^{2} + \eta^{2} + \xi^{2}\eta^{2} - 3\xi^{4}] \partial_{\xi} \psi \\ + \frac{1}{(\xi^{2}-1)} [\lambda + 6 - c^{2}\xi^{2} + \frac{(m^{2}+8)}{(\xi^{2}-1)}] \partial_{\xi} \psi \\ - \frac{\xi}{(\xi^{2}-1)} \{2c^{2} + \frac{4}{(\xi^{2}-1)} [\lambda - c^{2}\xi^{2}] + \frac{6m^{2}}{(\xi^{2}-1)^{2}} - \frac{3(1-\eta^{2})}{(\xi^{2}-\eta^{2})} [\lambda - c^{2}\xi^{2} + \frac{m^{2}}{(\xi^{2}-1)}] \} \psi \end{split}$$

$$\begin{split} \nabla_{\eta} \nabla_{\eta} \nabla_{\eta} \Psi &= + \frac{3\xi(\xi^{2}-1)}{(1-\eta^{2})(\xi^{2}-\eta^{2})} \partial_{\xi} \partial_{\eta} \Psi \\ &+ \frac{2\xi\eta(\xi^{2}-1)}{(1-\eta^{2})(\xi^{2}-\eta^{2})} \left[\frac{2}{(\xi^{2}-\eta^{2})} + \frac{1}{(1-\eta^{2})} - \frac{(\xi^{2}-1)}{(1-\eta^{2})(\xi^{2}-\eta^{2})} \right] \partial_{\xi} \Psi \\ &- \frac{3(\xi^{2}-1)}{(1-\eta^{2})(\xi^{2}-\eta^{2})^{2}} \left[\xi^{2} + \eta^{2} + \xi^{2}\eta^{2} - 3\eta^{4} \right] \partial_{\eta} \Psi \\ &- \frac{1}{(1-\eta^{2})} \left\{ \lambda + 6 - c^{2}\eta^{2} - \frac{(m^{2}+8)}{(1-\eta^{2})} \right\} \partial_{\eta} \Psi \\ &+ \frac{\eta}{(1-\eta^{2})} \left\{ 2c^{2} - \frac{4}{(1-\eta^{2})} \left[\lambda - c^{2}\eta^{2} \right] + \frac{6m^{2}}{(1-\eta^{2})^{2}} + \frac{3(\xi^{2}-1)}{(1-\eta^{2})(\xi^{2}-1)} \left[\lambda - c^{2}\eta^{2} - \frac{m^{2}}{(1-\eta^{2})} \right] \right\} \Psi \end{split}$$

.

$$\nabla_{\varphi}\nabla_{\varphi}\nabla_{\varphi}\Psi = +\frac{3(1-\eta^2)(\xi^2-1)}{(\xi^2-\eta^2)}[\xi\partial_{\xi}\partial_{\varphi}\psi - \eta\partial_{\eta}\partial_{\varphi}\psi] - (m^2+2)\partial_{\varphi}\psi$$

*

.

4

$$\begin{split} \nabla_{\xi} \nabla_{\xi} \nabla_{\eta} \psi &= -\xi \left[\frac{2}{(\xi^{2} - 1)} + \frac{2}{(\xi^{2} - \eta^{2})} - \frac{(1 - \eta^{2})}{(\xi^{2} - 1)(\xi^{2} - \eta^{2})} \right] \partial_{\xi} \partial_{\eta} \psi \\ &+ \frac{\eta}{(\xi^{2} - 1)(\xi^{2} - \eta^{2})} \left[3\lambda - 2c^{2}\xi^{2} - c^{2}\eta^{2} + \frac{2m^{2}}{(\xi^{2} - 1)} - \frac{m^{2}}{(1 - \eta^{2})} \right] \psi \\ &+ \frac{2\xi\eta}{(\xi^{2} - 1)(\xi^{2} - \eta^{2})^{2}} \left[3 - 4\xi^{2} + \eta^{2} \right] \partial_{\xi} \psi \\ &+ \frac{1}{(\xi^{2} - 1)} \left[\lambda - c^{2}\xi^{2} + \frac{m^{2}}{(\xi^{2} - 1)} \right] \partial_{\eta} \psi \\ &+ \frac{1}{(\xi^{2} - 1)(\xi^{2} - \eta^{2})^{2}} \left[3\eta^{4} - 3\eta^{2} + \xi^{2}\eta^{2} - 3\xi^{2} + 2\xi^{4} \right] \partial_{\eta} \psi \end{split}$$

$$\begin{split} \nabla_{\xi} \nabla_{\eta} \nabla_{\eta} \psi &= \eta [\frac{2}{(1-\eta^2)} + \frac{2}{(\xi^2 - \eta^2)} - \frac{(\xi^2 - 1)}{(1-\eta^2)(\xi^2 - \eta^2)}] \partial_{\xi} \partial_{\eta} \psi \\ &+ \frac{\xi}{(1-\eta^2)(\xi^2 - \eta^2)} \{ 3\lambda - 2c^2\eta^2 - c^2\xi^2 - \frac{2m^2}{(1-\eta^2)} + \frac{m^2}{(\xi^2 - 1)} \} \psi \\ &- \frac{2\xi\eta}{(1-\eta^2)(\xi^2 - \eta^2)^2} [3 - 4\eta^2 + \xi^2] \partial_{\eta} \psi \\ &- \frac{1}{(1-\eta^2)} [\lambda - c^2\eta^2 - \frac{m^2}{(1-\eta^2)}] \partial_{\xi} \psi \\ &- \frac{1}{(1-\eta^2)(\xi^2 - \eta^2)^2} [3\xi^4 - 3\xi^2 + \xi^2\eta^2 - 3\eta^2 + 2\eta^4] \partial_{\xi} \psi \end{split}$$

$$\begin{split} \nabla_{\xi} \nabla_{\eta} \nabla_{\varphi} \psi &= \partial_{\xi} \partial_{\eta} \partial_{\varphi} \psi + \left[\frac{\eta}{(1-\eta^{2})} + \frac{\eta}{(\xi^{2}-\eta^{2})} \right] \partial_{\xi} \partial_{\varphi} \psi - \left[\frac{\xi}{(\xi^{2}-1)} + \frac{\xi}{(\xi^{2}-\eta^{2})} \right] \partial_{\eta} \partial_{\varphi} \psi \\ &- \left[\frac{\xi\eta}{(1-\eta^{2})(\xi^{2}-\eta^{2})} + \frac{\xi\eta}{(1-\eta^{2})(\xi^{2}-1)} + \frac{\xi\eta}{(\xi^{2}-1)(\xi^{2}-\eta^{2})} \right] \partial_{\varphi} \psi \\ \nabla_{\xi} \nabla_{\xi} \nabla_{\varphi} \psi &= + \frac{(1-\eta^{2})}{(\xi^{2}-1)(\xi^{2}-\eta^{2})} \left[\xi \partial_{\xi} \partial_{\varphi} \psi - \eta \partial_{\eta} \partial_{\varphi} \psi \right] \\ &+ \frac{1}{(\xi^{2}-1)} \left[\lambda + 2 - c^{2} \xi^{2} + \frac{m^{2}+2}{(\xi^{2}-1)} \right] \partial_{\varphi} \psi - \frac{4}{(\xi^{2}-1)} \xi \partial_{\xi} \partial_{\varphi} \psi \\ \nabla_{\eta} \nabla_{\eta} \nabla_{\varphi} \psi &= + \frac{(\xi^{2}-1)}{(1-\eta^{2})(\xi^{2}-\eta^{2})} \left[\xi \partial_{\xi} \partial_{\varphi} \psi - \eta \partial_{\eta} \partial_{\varphi} \psi \right] \\ &- \frac{1}{(1-\eta^{2})} \left[\lambda + 2 - c^{2} \eta^{2} - \frac{m^{2}+2}{(1-\eta^{2})} \right] \partial_{\varphi} \psi + \frac{4}{(1-\eta^{2})} \eta \partial_{\eta} \partial_{\varphi} \psi \\ \nabla_{\xi} \nabla_{\varphi} \nabla_{\varphi} \psi &= - \frac{\eta (1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \partial_{\xi} \partial_{\eta} \psi \\ &+ \frac{2\xi\eta (1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})^{2}} \partial_{\eta} \psi - \frac{2\xi^{2}(1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \partial_{\xi} \psi - \left[m^{2} + \frac{(1-\eta^{2})(\xi^{2}+1)}{(\xi^{2}-\eta^{2})} \right] \partial_{\xi} \psi \\ &+ \frac{\xi(1-\eta^{2})}{(\xi^{2}-\eta^{2})} \left[\lambda - c^{2} \xi^{2} + \frac{m^{2}}{(\xi^{2}-1)} \right] \psi + \frac{2\xi m^{2}}{(\xi^{2}-1)} \psi \\ \nabla_{\eta} \nabla_{\varphi} \nabla_{\varphi} \psi &= + \frac{\xi(1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \partial_{\xi} \partial_{\eta} \psi \\ &+ \frac{2\xi\eta (1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \partial_{\xi} \psi - \frac{2\eta^{2}(1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \partial_{\eta} \psi - \left\{ m^{2} + \frac{(1+\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \right\} \partial_{\eta} \psi \\ &+ \frac{2\xi\eta (1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \partial_{\xi} \psi - \frac{2\eta^{2}(1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \partial_{\eta} \psi - \left\{ m^{2} + \frac{(1+\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \right\} \partial_{\eta} \psi \\ &+ \frac{2\xi\eta (1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \partial_{\xi} \psi - \frac{2\eta^{2}(1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \partial_{\eta} \psi - \left\{ m^{2} + \frac{(1+\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \right\} \partial_{\eta} \psi \\ &+ \frac{2\xi\eta (1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \partial_{\xi} \psi - \frac{2\eta^{2}(1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \partial_{\eta} \psi - \left\{ m^{2} + \frac{(1+\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \right\} \partial_{\eta} \psi \\ &+ \frac{2\xi\eta (1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \partial_{\xi} \psi - \frac{2\eta^{2}(1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \partial_{\eta} \psi - \left\{ m^{2} + \frac{(1+\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \right\} \partial_{\eta} \psi \\ &+ \frac{2\xi\eta (1-\eta^{2})(\xi^{2}-1)}{(\xi^{2}-\eta^{2})} \partial_{\xi} \psi \\ &+$$

$$+\frac{2\zeta\eta(1-\eta^{2})(\zeta-1)}{(\xi^{2}-\eta^{2})^{2}}\partial_{\xi}\psi - \frac{2\eta(1-\eta^{2})(\zeta-1)}{(\xi^{2}-\eta^{2})^{2}}\partial_{\eta}\psi - \{m^{2} + \frac{(1+\eta^{2})(\zeta-1)}{(\xi^{2}-\eta^{2})}\}\partial_{\eta}\psi + \frac{\eta(\xi^{2}-1)}{(\xi^{2}-\eta^{2})}[\lambda - c^{2}\eta^{2} - \frac{m^{2}}{(1-\eta^{2})}]\psi - \frac{2\eta m^{2}}{(1-\eta^{2})}\psi$$

G-6

The component expression for the above vectors in terms of the scalar basis functions are as follows:

$$\begin{split} \sqrt{\lambda_n} (V_n^1)_{\xi} &= +f \frac{\eta}{(\xi^2 - 1)} \partial_{\varphi} \psi_n^T \\ \sqrt{\lambda_n} (V_n^1)_{\eta} &= -f \frac{\xi}{(1 - \eta^2)} \partial_{\varphi} \psi_n^T \\ \sqrt{\lambda_n} (V_n^1)_{\varphi} &= +f \frac{(1 - \eta^2)(\xi^2 - 1)}{(\xi^2 - \eta^2)} \{\xi \partial_{\eta} \psi_n^T - \eta \partial_{\xi} \psi_n^T \} \\ k_T \sqrt{\lambda_n} (V_n^2)_{\xi} &= c_T^2 \xi \psi_n^T + 2 \nabla_{\xi} \psi_n^T + \frac{\xi(\xi^2 \mp 1)}{(\xi^2 \mp \eta^2)} \nabla_{\xi} \nabla_{\xi} \psi_n^T + \frac{\eta(1 - \eta^2)}{(\xi^2 \mp \eta^2)} \nabla_{\xi} \nabla_{\eta} \psi_n^T \\ k_T \sqrt{\lambda_n} (V_n^2)_{\eta} &= c_T^2 \eta \psi_n^T + 2 \nabla_{\eta} \psi_n^T + \frac{\xi(\xi^2 \mp 1)}{(\xi^2 \mp \eta^2)} \nabla_{\xi} \nabla_{\eta} \psi_n^T + \frac{\eta(1 - \eta^2)}{(\xi^2 \mp \eta^2)} \nabla_{\eta} \nabla_{\eta} \psi_n^T \\ k_T \sqrt{\lambda_n} (V_n^2)_{\varphi} &= 2 \nabla_{\varphi} \psi_n^T + \frac{\xi(\xi^2 \mp 1)}{(\xi^2 \mp \eta^2)} \nabla_{\xi} \nabla_{\varphi} \psi_n^T + \frac{\eta(1 - \eta^2)}{(\xi^2 \mp \eta^2)} \nabla_{\eta} \nabla_{\varphi} \psi_n^T \end{split}$$

$$(V_n^3)_{\xi} = \frac{f}{c_L} \partial_{\xi} \psi_n^L$$
$$(V_n^3)_{\eta} = \frac{f}{c_L} \partial_{\eta} \psi_n^L$$
$$(V_n^3)_{\varphi} = \frac{f}{c_L} \partial_{\varphi} \psi_n^L$$

The components of the covariant derivatives of the vector basis functions are as follows:

$$\frac{1}{f}\sqrt{\lambda}\nabla_{\xi}(V_{n}^{1})_{\xi} = +\eta(1-\eta^{2})\nabla_{\xi}\nabla_{\varphi}\psi_{n}^{T}$$
$$\frac{1}{f}\sqrt{\lambda}\nabla_{\eta}(V_{n}^{1})_{\eta} = -\xi(\xi^{2}-1)\nabla_{\eta}\nabla_{\varphi}\psi_{n}^{T}$$

$$\frac{1}{f}\sqrt{\lambda}\nabla_{\xi}(V_{n}^{1})_{\eta} = -\frac{\xi}{(1-\eta^{2})}\nabla_{\xi}\nabla_{\varphi}\psi_{n}^{T} - \frac{(\xi^{2}-\eta^{2})}{(\xi^{2}-1)(1-\eta^{2})}\nabla_{\varphi}\psi_{n}^{T}$$
$$\frac{1}{f}\sqrt{\lambda}\nabla_{\eta}(V_{n}^{1})_{\xi} = +\frac{\eta}{(\xi^{2}-1)}\nabla_{\eta}\nabla_{\varphi}\psi_{n}^{T} + \frac{(\xi^{2}-\eta^{2})}{(\xi^{2}-1)(1-\eta^{2})}\nabla_{\varphi}\psi_{n}^{T}$$

$$\frac{1}{f}\sqrt{\lambda}\nabla_{\xi}(V_n^1)_{\varphi} = +(1-\eta^2)\nabla_{\eta}\psi_n^T - \frac{(\xi^2-1)(1-\eta^2)}{(\xi^2-\eta^2)}[\eta\nabla_{\xi}\nabla_{\xi}\psi_n^T - \xi\nabla_{\xi}\nabla_{\eta}\psi_n^T]$$
$$\frac{1}{f}\sqrt{\lambda}\nabla_{\eta}(V_n^1)_{\varphi} = -(\xi^2-1)\nabla_{\xi}\psi_n^T - \frac{(\xi^2-1)(1-\eta^2)}{(\xi^2-\eta^2)}[\eta\nabla_{\xi}\nabla_{\eta}\psi_n^T - \xi\nabla_{\eta}\nabla_{\eta}\psi_n^T]$$

$$\frac{1}{f}\sqrt{\lambda}\nabla_{\varphi}(V_{n}^{1})_{\xi} = -(1-\eta^{2})\nabla_{\eta}\psi_{n}^{T} + \frac{\eta}{(\xi^{2}-1)}\nabla_{\varphi}\nabla_{\varphi}\psi_{n}^{T}$$
$$\frac{1}{f}\sqrt{\lambda}\nabla_{\varphi}(V_{n}^{1})_{\eta} = +(\xi^{2}-1)\nabla_{\xi}\psi_{n}^{T} - \frac{\xi}{(1-\eta^{2})}\nabla_{\varphi}\nabla_{\varphi}\psi_{n}^{T}$$

$$\frac{1}{f}\sqrt{\lambda}\nabla_{\varphi}(V_n^1)_{\varphi} = -\frac{(\xi^2 - 1)(1 - \eta^2)}{(\xi^2 - \eta^2)} [\eta\nabla_{\xi}\nabla_{\varphi}\psi_n^T - \xi\nabla_{\eta}\nabla_{\varphi}\psi_n^T]$$

$$\begin{split} k_{T}\sqrt{\lambda_{n}}\nabla_{\xi}(V_{n}^{2})_{\xi} &= c_{T}^{2}\frac{(\xi^{2}-\eta^{2})}{(\xi^{2}-1)}\psi_{n}^{T} + c_{T}^{2}\xi\nabla_{\xi}\psi_{n}^{T} + 3\nabla_{\xi}\nabla_{\xi}\psi_{n}^{T} + \frac{\xi(\xi^{2}-1)}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\xi}\nabla_{\xi}\psi_{n}^{T} + \frac{\eta(1-\eta^{2})}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\xi}\nabla_{\eta}\psi_{n}^{T} \\ k_{T}\sqrt{\lambda_{n}}\nabla_{\eta}(V_{n}^{2})_{\eta} &= c_{T}^{2}\frac{(\xi^{2}-\eta^{2})}{(1-\eta^{2})}\psi_{n}^{T} + c_{T}^{2}\eta\nabla_{\eta}\psi_{n}^{T} + 3\nabla_{\eta}\nabla_{\eta}\psi_{n}^{T} + \frac{\xi(\xi^{2}-1)}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\eta}\nabla_{\eta}\psi_{n}^{T} + \frac{\eta(1-\eta^{2})}{(\xi^{2}-\eta^{2})}\nabla_{\eta}\nabla_{\eta}\nabla_{\eta}\nabla_{\eta}\psi_{n}^{T} \\ k_{T}\sqrt{\lambda_{n}}\nabla_{\varphi}(V_{n}^{2})_{\varphi} &= c_{T}^{2}(\xi^{2}-1)(1-\eta^{2})\psi_{n}^{T} + 3\nabla_{\varphi}\nabla_{\varphi}\psi_{n}^{T} + \frac{\xi(\xi^{2}-1)}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\varphi}\nabla_{\varphi}\psi_{n}^{T} + \frac{\eta(1-\eta^{2})}{(\xi^{2}-\eta^{2})}\nabla_{\varphi}\nabla_{\varphi}\nabla_{\eta}\psi_{n}^{T} \\ k_{T}\sqrt{\lambda_{n}}\nabla_{\varphi}(V_{n}^{2})_{\varphi} &= c_{T}^{2}\eta\nabla_{\xi}\psi_{n}^{T} + 3\nabla_{\xi}\nabla_{\eta}\psi_{n}^{T} + \frac{\xi(\xi^{2}-1)}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\varphi}\nabla_{\eta}\psi_{n}^{T} + \frac{\eta(1-\eta^{2})}{(\xi^{2}-\eta^{2})}\nabla_{\varphi}\nabla_{\varphi}\nabla_{\eta}\psi_{n}^{T} \\ k_{T}\sqrt{\lambda_{n}}\nabla_{\xi}(V_{n}^{2})_{\varphi} &= c_{T}^{2}\eta\nabla_{\xi}\psi_{n}^{T} + 3\nabla_{\xi}\nabla_{\eta}\psi_{n}^{T} + \frac{\xi(\xi^{2}-1)}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\varphi}\nabla_{\eta}\psi_{n}^{T} + \frac{\eta(1-\eta^{2})}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\eta}\nabla_{\eta}\psi_{n}^{T} \\ k_{T}\sqrt{\lambda_{n}}\nabla_{\eta}(V_{n}^{2})_{\xi} &= c_{T}^{2}\xi\nabla_{\eta}\psi_{n}^{T} + 3\nabla_{\xi}\nabla_{\eta}\psi_{n}^{T} + \frac{\xi(\xi^{2}-1)}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\xi}\nabla_{\eta}\psi_{n}^{T} + \frac{\eta(1-\eta^{2})}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\eta}\nabla_{\eta}\psi_{n}^{T} \\ k_{T}\sqrt{\lambda_{n}}\nabla_{\eta}(V_{n}^{2})_{\varphi} &= 3\nabla_{\xi}\nabla_{\varphi}\psi_{n}^{T} + \frac{\xi(\xi^{2}-1)}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\xi}\nabla_{\varphi}\psi_{n}^{T} + \frac{\eta(1-\eta^{2})}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\eta}\nabla_{\eta}\psi_{n}^{T} \\ k_{T}\sqrt{\lambda_{n}}\nabla_{\eta}(V_{n}^{2})_{\xi} &= c_{T}^{2}\xi\nabla_{\varphi}\psi_{n}^{T} + \frac{\xi(\xi^{2}-1)}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\varphi}\psi_{n}^{T} + \frac{\eta(1-\eta^{2})}{(\xi^{2}-\eta^{2})}\nabla_{\chi}\nabla_{\eta}\nabla_{\eta}\psi_{n}^{T} \\ k_{T}\sqrt{\lambda_{n}}}\nabla_{\varphi}(V_{n}^{2})_{\xi} &= c_{T}^{2}\xi\nabla_{\varphi}\psi_{n}^{T} + 3\nabla_{\eta}\nabla_{\varphi}\psi_{n}^{T} + \frac{\xi(\xi^{2}-1)}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\varphi}\psi_{\eta}\psi_{n}^{T} + \frac{\eta(1-\eta^{2})}{(\xi^{2}-\eta^{2})}\nabla_{\varphi}\nabla_{\eta}\nabla_{\eta}\psi_{n}^{T} \\ k_{T}\sqrt{\lambda_{n}}}\nabla_{\varphi}(V_{n}^{2})_{\xi} &= c_{T}^{2}\xi\nabla_{\varphi}\psi_{n}^{T} + 3\nabla_{\xi}\nabla_{\varphi}\psi_{n}^{T} + \frac{\xi(\xi^{2}-1)}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\varphi}\psi_{n}^{T} + \frac{\eta(1-\eta^{2})}{(\xi^{2}-\eta^{2})}\nabla_{\xi}\nabla_{\eta}\nabla_{\eta}\psi_{n}^{T} \\ k_{T}\sqrt{\lambda_{n}}\nabla_{\varphi$$

G-9

u

$$\nabla_{\xi}(V_{n}^{3})_{\xi} = \frac{f}{c_{L}}\nabla_{\xi}\nabla_{\xi}\psi_{n}^{L}$$

$$\nabla_{\eta}(V_{n}^{3})_{\eta} = \frac{f}{c_{L}}\nabla_{\eta}\nabla_{\eta}\psi_{n}^{L}$$

$$\nabla_{\varphi}(V_{n}^{3})_{\varphi} = \frac{f}{c_{L}}\nabla_{\varphi}\nabla_{\varphi}\psi_{n}^{L}$$

$$\nabla_{\xi}(V_{n}^{3})_{\eta} = \frac{f}{c_{L}}\nabla_{\xi}\nabla_{\eta}\psi_{n}^{L}$$

$$\nabla_{\eta}(V_{n}^{3})_{\xi} = \frac{f}{c_{L}}\nabla_{\xi}\nabla_{\eta}\psi_{n}^{L}$$

$$\nabla_{\xi}(V_{n}^{3})_{\varphi} = \frac{f}{c_{L}}\nabla_{\xi}\nabla_{\varphi}\psi_{n}^{L}$$

$$\nabla_{\varphi}(V_{n}^{3})_{\xi} = \frac{f}{c_{L}}\nabla_{\xi}\nabla_{\varphi}\psi_{n}^{L}$$

$$\nabla_{\eta}(V_{n}^{3})_{\varphi} = \frac{f}{c_{L}}\nabla_{\eta}\nabla_{\varphi}\psi_{n}^{L}$$

$$\nabla_{\varphi}(V_{n}^{3})_{\eta} = \frac{f}{c_{L}}\nabla_{\eta}\nabla_{\varphi}\psi_{n}^{L}$$

.

APPENDIX H

PARITY OF BASIS FUNCTIONS

Approved for public release; distribution is unlimited

APPENDIX H - PARITY OF BASIS FUNCTIONS

This appendix tabulates the parity of the various functions and their derivatives with respect to the transformation $\eta \rightarrow -\eta$, and the dependence of the functions on the first order derivative with respect to the azimuthal angle. These symmetries are important for determining the non-zero matrix elements obtained from the surface integrals for a symmetric target such as a prolate spheroid or finite cylinder.

Table H-1 contains the polar and azimuthal parity of the vector harmonics. Table H-2 contains the parity of the components of the vector basis functions, and Table H-3 contains the parity of the stress tensor formed from the covariant derivative of the vector basis functions. The interpretation of the polar parity column is as follows: +1 signifies the function transforms as $F(-\eta) = +(-1)^{l-m} F(\eta)$, -1 signifies the function transforms as $F(-\eta) = -(-1)^{l-m} F(\eta)$, and 0 signifies the function is zero. The interpretation of the azimuthal parity column is as follows: Y signifies the function is proportional to an odd number of derivatives of the function with respect to the azimuthal order, and N signifies it is proportional to an even number of derivatives.

The significance of the azimuthal parity is that it denotes the coupling of different parities in the azimuthal integration. The significance of the polar parity term is that it dictates the values of order *l* for which the integrals over the coordinate $-1 < \eta < +1$ are non-zero.

VECTOR HARMONIC	COMPONENT	POLAR PARITY	AZIMUTHAL PARITY		
First Vector					
1	Ę.	0	0		
1	η	+1	Y		
1	φ	-1	N		
Second Vector					
2	ξ	0	0		
2	η	-1	N		
2	φ	+1	Y		
Third Vector					
3	ξ	+1	N		
3	η	0	0		
3	φ	0	0		

TABLE H-1. PARITY OF VEC	CTOR HARMONICS
---------------------------------	-----------------------

VECTOR BASIS FUNCTION	COMPONENT	POLAR PARITY	AZIMUTHAL PARITY			
First Vector						
1	ξ	-1	Y			
1	η	+1	Y			
1	φ	-1	N			
Second Vector						
2	ξ	+1	N			
2	η	-1	N			
2	φ	+1	Υ			
Third Vector						
3	ξ	+1	Ν			
3	η	-1	N			
3	φ	+	Y			

TABLE H-2. PARITY OF VECTOR BASIS FUNCTIONS

TABLE H-3. PARITY OF STRESS TENSOR FORMED FROM VECTOR BASIS FUNCTIONS

VECTOR BASIS	COMPONENT OF STRESS	POLAR	AZIMUTHAL				
FUNCTION TENSOR PARILY PARILY First Basis Function First Basis Function First Basis Function First Basis Function							
1	(ξ,ξ)	-1	Y				
1	$(\xi,\eta)(\eta,\xi)$	+1	Y				
1	$(\xi,\varphi)(\varphi,\xi)$	-1	N				
1	(η,η)	-1	Y				
1	$(\eta, \varphi)(\varphi, \eta)$	+1	N				
1	(φ, φ)	-1	Y				
Second Basis Function							
2	(5,5)	+1	N				
2	$(\xi,\eta)(\eta,\xi)$	-1	N				
2	$(\xi, \varphi)(\varphi, \xi)$	+1	Y				
2	(η,η)	+1	N				
2	$(\eta, \varphi)(\varphi, \eta)$	-1	Y				
2	(φ, φ)	+1	N				
Third Basis Function							
3	(5,5)	+1	N				
3	$(\xi,\eta)(\eta,\xi)$	-1	N				
3	$(\xi, \varphi)(\varphi, \xi)$	+1	Y				
3	(η,η)	+1	N				
3	$(\eta, \varphi)(\varphi, \eta)$	-1	Y				
3	(φ, φ)	+1	N				

APPENDIX I

COMPUTATION OF T-MATRIX FOR PROLATE SPHEROID

Approved for public release; distribution is unlimited

APPENDIX I - COMPUTATION OF T-MATRIX FOR PROLATE SPHEROID

This appendix describes the method for performing the surface integrals necessary to compute the spheroidal T-matrix for a prolate spheroid. Since the spheroid is both symmetric under the transformation $\eta \rightarrow -\eta$ and $\varphi \rightarrow \varphi + \alpha$, the T-matrix is diagonal in parity σ and azimuthal order *m*. In addition, only terms for which the difference (l-l') is even have non-zero elements for the T-matrix. These symmetries allow us to reduce the two-dimensional surface integrals to a one-dimensional integral over the variable $0 > \eta > 1$. Due to the above symmetries, the T-matrix is of the following form:

$$T_{\sigma m l, \sigma' m' l'} = \delta_{\sigma}^{\sigma'} \delta_{m}^{m'} T_{l, l'}^{m}$$

Define the following basis functions by replacing the trigonometric functions in the basis functions by the exponential functions $\exp[\pm im\varphi]$, respectively.

$$S^{W}_{\pm ml} \exp[\pm im\varphi] = \frac{1}{\sqrt{\varepsilon(m)}} (S^{W}_{eml} \pm iS^{W}_{oml})$$

$$A^{W}_{\pm ml} \exp[\pm im\varphi] = \frac{1}{\sqrt{\varepsilon(m)}} (A^{W}_{eml} \pm iA^{W}_{oml})$$

$$V^{W}_{\pm ml} \exp[\pm im\varphi] = \frac{1}{\sqrt{\varepsilon(m)}} (V^{W}_{eml} \pm iV^{W}_{oml})$$

$$A^{\tau}_{\pm ml} \exp[\pm im\varphi] = \frac{1}{\sqrt{\varepsilon(m)}} (A^{\tau}_{eml} \pm iA^{\tau}_{oml})$$

$$V^{\tau}_{\pm ml} \exp[\pm im\varphi] = \frac{1}{\sqrt{\varepsilon(m)}} (V^{\tau}_{eml} \pm iV^{\tau}_{oml})$$

$$t^{W}_{\pm ml} \exp[\pm im\varphi] = \frac{1}{\sqrt{\varepsilon(m)}} (t(V^{w}_{eml}) \pm it(V^{W}_{oml}))$$

$$t^{\tau}_{\pm ml} \exp[\pm im\varphi] = \frac{1}{\sqrt{\varepsilon(m)}} (t(V^{w}_{eml}) \pm it(V^{W}_{oml}))$$

The functions S^{W} , A^{W} , V^{W} , and t^{W} , are the scalar harmonic, vector harmonic, vector basis function, and traction vector in the external fluid, respectively. The functions A^{τ} , V^{τ} , and t^{τ} are the vector harmonics, vector basis functions, and traction vectors in the interior, respectively. Depending upon whether a given function $F_{\pm ml}$ contains an even or odd number of derivatives with respect to the angle φ the function will be proportional to either "1" or $\pm i$. Define the following notation:

 $F_{\pm ml} = \tilde{F}_{ml} \begin{cases} \pm 1, \text{ even azimuthal parity} \\ \pm i, \text{ odd azimuthal parity} \end{cases}$

Define the following functions based on their azimuthal parity:

$$\begin{split} \tilde{S}_{ml}^{W} &= S_{\pm ml}^{W} \\ (\tilde{A}_{ml}^{1})_{\eta} &= (A_{\pm ml}^{1})_{\eta} / \pm i \\ (\tilde{A}_{ml}^{1})_{\varphi} &= (A_{\pm ml}^{1})_{\varphi} \\ (\tilde{A}_{ml}^{2})_{\eta} &= (A_{\pm ml}^{2})_{\eta} \\ (\tilde{A}_{ml}^{2})_{\varphi} &= (A_{\pm ml}^{2})_{\varphi} / \pm i \\ (\tilde{A}_{ml}^{3})_{\xi} &= (A_{\pm ml}^{3})_{\xi} \end{split}$$

$$\begin{split} (\tilde{V}_{ml}^{1})_{\xi} &= (V_{\pm ml}^{1})_{\xi} / \pm i \\ (\tilde{V}_{ml}^{1})_{\eta} &= (V_{\pm ml}^{1})_{\eta} / \pm i \\ (\tilde{V}_{ml}^{1})_{\varphi} &= (V_{\pm ml}^{1})_{\varphi} \\ (\tilde{V}_{ml}^{2})_{\xi} &= (V_{\pm ml}^{2})_{\xi} \\ (\tilde{V}_{ml}^{2})_{\eta} &= (V_{\pm ml}^{2})_{\eta} \\ (\tilde{V}_{ml}^{2})_{\varphi} &= (V_{\pm ml}^{2})_{\varphi} / \pm i \\ (\tilde{V}_{ml}^{3})_{\xi} &= (V_{\pm ml}^{3})_{\xi} \\ (\tilde{V}_{ml}^{3})_{\eta} &= (V_{\pm ml}^{3})_{\eta} \\ (\tilde{V}_{ml}^{3})_{\eta} &= (V_{\pm ml}^{3})_{\eta} / \pm i \end{split}$$

$$\begin{split} (\tilde{t}_{ml}^{1})_{\xi} &= (t_{\pm ml}^{1})_{\xi} / \pm i \\ (\tilde{t}_{ml}^{1})_{\eta} &= (t_{\pm ml}^{1})_{\eta} / \pm i \\ (\tilde{t}_{ml}^{1})_{\varphi} &= (t_{\pm ml}^{1})_{\varphi} \\ (\tilde{t}_{ml}^{2})_{\xi} &= (t_{\pm ml}^{2})_{\xi} \\ (\tilde{t}_{ml}^{2})_{\eta} &= (t_{\pm ml}^{2})_{\eta} \\ (\tilde{t}_{ml}^{2})_{\varphi} &= (t_{\pm ml}^{2})_{\varphi} / \pm i \\ (\tilde{t}_{ml}^{3})_{\xi} &= (t_{\pm ml}^{3})_{\xi} \\ (\tilde{t}_{ml}^{3})_{\eta} &= (t_{\pm ml}^{3})_{\eta} \\ (\tilde{t}_{ml}^{3})_{\varphi} &= (t_{\pm ml}^{3})_{\varphi} / \pm i \end{split}$$

The computation of the T-matrix requires the evaluation of the following surface integrals over the surface of the target:

$$Q_{n,\tau'n'} = -i\frac{k}{\lambda^{W}} \int_{S} dA\{(n \bullet t(V_n^W)(n \bullet A_{\tau',n'}))\}$$

$$M_{n,n'} = -i\frac{k}{\lambda^{W}}\int_{S} dA\{(n \bullet V_{n}^{W})S_{n'}^{W}\}$$

$$P_{\tau n,n'} = \int_{S} dA\{(n \bullet \operatorname{Re} V_{\tau,n}) S^{W}_{n'}\}$$

$$R_{\tau n,\tau'n'} = \int_{S} dA\{t(\operatorname{Re} V_{\tau n}) \bullet A_{\tau'n'}\}$$

In the above integrals one adopts the convention that the left hand term in the integral has +m and the right hand term has -m for the azimuthal angle dependence of the integrand. The above integrals are diagonal in m, and they can be replaced by the following one-dimensional integrals:

$$h_{\xi} = f \sqrt{(\xi^{2} - \eta^{2})/(\xi^{2} - 1)}$$

$$h_{\eta} = f \sqrt{(\xi^{2} - \eta^{2})/(1 - \eta^{2})}$$

$$h_{\varphi} = f \sqrt{(\xi^{2} - 1)(1 - \eta^{2})}$$

$$\partial \xi(\eta)$$

$$\Delta_n = h_\eta h_\varphi n_{\xi} - h_{\xi} h_\varphi n_{\eta} \frac{\partial \xi(\eta)}{\partial \eta}$$

$$Q^{m}_{l,\tau'l'} = -2\pi i \frac{k}{\lambda^{W}} \int_{-1}^{+1} d\eta \Delta_{n} (t^{W}_{+ml})_{\xi} (A^{\tau'}_{-ml'})_{\xi}$$
$$M^{m}_{l,l'} = -2\pi i \frac{k}{\lambda^{W}} \int_{-1}^{+1} d\eta \Delta_{n} (V^{W}_{+ml})_{\xi} (S^{W}_{-ml'})$$
$$P^{m}_{\tau l,l'} = 2\pi \int_{-1}^{+1} d\eta \Delta_{n} (\operatorname{Re} V^{\tau}_{+ml})_{\xi} (S^{W}_{-ml'})$$

$$R^{m}_{\tau l,\tau' l'} = 2\pi \int_{-1}^{\infty} d\eta \Delta_n (\operatorname{Re} t^{\tau}_{+ml}) \bullet (A^{\tau'}_{-ml'})$$

Using the polar parity and the azimuthal parity properties of the integrands, the non-zero components of the above integrals in the case l-l' is even are given by the following integrals:

$$\begin{split} & Q^{m}_{l,3l'} = -4\pi i \frac{k}{\lambda^{W}} \int_{0}^{+1} d\eta \Delta_{n}(\tilde{t}_{ml}^{W})_{\xi}(\tilde{A}_{ml'}^{3})_{\xi} \\ & M^{m}_{l,l'} = -4\pi i \frac{k}{\lambda^{W}} \int_{0}^{+1} d\eta \Delta_{n}(\tilde{V}_{ml}^{W})_{\xi}(\tilde{S}_{ml'}^{W}) \\ & P^{m}_{2l,l'} = 4\pi \int_{0}^{+1} d\eta \Delta_{n}(\operatorname{Re}\tilde{V}_{ml}^{2})_{\xi}(\tilde{S}_{ml'}^{W}) \\ & P^{m}_{3l,l'} = 4\pi \int_{0}^{+1} d\eta \Delta_{n}(\operatorname{Re}\tilde{V}_{ml}^{3})_{\xi}(\tilde{S}_{ml'}^{W}) \\ & R^{m}_{1l,1l'} = 4\pi \int_{0}^{+1} d\eta \Delta_{n}[+(\operatorname{Re}\tilde{t}_{ml}^{1})_{\eta}(\tilde{A}_{ml'}^{1})_{\eta} + (\operatorname{Re}\tilde{t}_{ml}^{1})_{\phi}(\tilde{A}_{ml'}^{1})_{\phi}] \\ & R^{m}_{2l,2l'} = 4\pi \int_{0}^{+1} d\eta \Delta_{n}[+(\operatorname{Re}\tilde{t}_{ml}^{2})_{\eta}(\tilde{A}_{ml'}^{2})_{\eta} + (\operatorname{Re}\tilde{t}_{ml}^{2})_{\phi}(\tilde{A}_{ml'}^{2})_{\phi}] \\ & R^{m}_{3l,3l'} = 4\pi \int_{0}^{+1} d\eta \Delta_{n}(\operatorname{Re}\tilde{t}_{ml}^{3})_{\xi}(\tilde{A}_{ml'}^{3})_{\xi} \\ & R^{m}_{2l,3l'} = 4\pi \int_{0}^{+1} d\eta \Delta_{n}(\operatorname{Re}\tilde{t}_{ml}^{2})_{\xi}(\tilde{A}_{ml'}^{3})_{\xi} \\ & R^{m}_{3l,2l'} = 4\pi \int_{0}^{+1} d\eta \Delta_{n}[+(\operatorname{Re}\tilde{t}_{ml}^{3})_{\eta}(\tilde{A}_{ml'}^{2})_{\eta} + (\operatorname{Re}\tilde{t}_{ml}^{3})_{\phi}(\tilde{A}_{ml'}^{2})_{\phi}] \end{split}$$

The non-zero components of the integrals in the case l-l' is odd are given by the following integrals:

$$P^{m}_{1l,l'} = +4\pi i \int_{0}^{+1} d\eta \Delta_{n} (\operatorname{Re} \tilde{V}^{1}_{ml})_{\xi} (\tilde{S}^{W}_{ml'})$$

$$R^{m}_{1l,2l'} = +4\pi i \int_{0}^{+1} d\eta \Delta_{n} [+ (\operatorname{Re} \tilde{t}^{1}_{ml})_{\eta} (\tilde{A}^{2}_{ml'})_{\eta} - (\operatorname{Re} \tilde{t}^{1}_{ml})_{\psi} (\tilde{A}^{2}_{ml'})_{\psi}]$$

$$R^{m}_{2l,1l'} = -4\pi i \int_{0}^{+1} d\eta \Delta_{n} [+ (\operatorname{Re} \tilde{t}^{2}_{ml})_{\eta} (\tilde{A}^{1}_{ml'})_{\eta} - (\operatorname{Re} \tilde{t}^{2}_{ml})_{\psi} (\tilde{A}^{1}_{ml'})_{\psi}]$$

$$R^{m}_{3l,1l'} = -4\pi i \int_{0}^{+1} d\eta \Delta_{n} [+ (\operatorname{Re} \tilde{t}^{3}_{ml})_{\hat{\eta}} (\tilde{A}^{1}_{ml'})_{\hat{\eta}} - (\operatorname{Re} \tilde{t}^{3}_{ml})_{\hat{\varphi}} (\tilde{A}^{1}_{ml'})_{\hat{\varphi}}]$$

$$R^{m}_{1l,3l'} = +i4\pi \int_{0}^{+1} d\eta \Delta_{n} (\operatorname{Re} \tilde{t}^{1}_{ml})_{\xi} (\tilde{A}^{3}_{ml'})_{\xi}$$

In the case (l-l') is even, the components of the *R* and *P* matrices are real. In the case (l-l') is odd the *R* and *P* matrices are imaginary. The *R* and *P* matrices can be transformed into real matrices by making the following similarity transformation on the vector indices of the *R*, *P*, and *Q* matrices:

$$\begin{split} S_{\tau\tau'} &= -i\delta_{\tau}^{\tau'}\delta_{\tau}^{1} + \delta_{\tau}^{\tau'}\delta_{\tau}^{2} + \delta_{\tau}^{\tau'}\delta_{\tau}^{3} \\ S^{-1}_{\tau\tau'} &= S^{\dagger}_{\tau\tau'} = +i\delta_{\tau}^{\tau'}\delta_{\tau}^{1} + \delta_{\tau}^{\tau'}\delta_{\tau}^{2} + \delta_{\tau}^{\tau'}\delta_{\tau}^{3} \\ R^{m}_{\tau l,\tau' l'} &= \sum_{\tau^{*},\tau^{*}} S_{\tau',\tau^{*}}R^{m}_{\tau^{*}l,\tau^{*}l''}S^{\dagger}_{\tau^{*},\tau'} \\ P^{m}_{\tau l,l'} &= \sum_{\tau^{*}} S_{\tau',\tau^{*}}P^{m}_{\tau^{*}l,l'} \\ Q^{m}_{l,\tau' l'} &= \sum_{\tau^{*}} Q^{m}_{l,\tau^{*}l''}S^{\dagger}_{\tau^{*},\tau'} \end{split}$$

In the case (l-l') is even, the components of the Q, R, and P matrices are left invariant by this transformation, whereas in the case (l-l') is odd they are transformed to the following real matrices:

$$P^{m}_{1l,l'} = +4\pi \int_{0}^{+1} d\eta \Delta_{n} (\operatorname{Re} \tilde{V}^{1}_{ml})_{\xi} (\tilde{S}^{W}_{ml'})$$

$$R^{m}_{1l,2l'} = +4\pi \int_{0}^{+1} d\eta \Delta_{n} [+ (\operatorname{Re} \tilde{t}^{1}_{ml})_{\eta} (\tilde{A}^{2}_{ml'})_{\eta} - (\operatorname{Re} \tilde{t}^{1}_{ml})_{\phi} (\tilde{A}^{2}_{ml'})_{\phi}]$$

$$R^{m}_{2l,1l'} = +4\pi \int_{0}^{+1} d\eta \Delta_{n} [+ (\operatorname{Re} \tilde{t}^{2}_{ml})_{\eta} (\tilde{A}^{1}_{ml'})_{\eta} - (\operatorname{Re} \tilde{t}^{2}_{ml})_{\phi} (\tilde{A}^{1}_{ml'})_{\phi}]$$

$$R^{m}_{3l,1l'} = +4\pi \int_{0}^{+1} d\eta \Delta_{n} [+ (\operatorname{Re} \tilde{t}^{3}_{ml})_{\hat{\eta}} (\tilde{A}^{1}_{ml'})_{\hat{\eta}} - (\operatorname{Re} \tilde{t}^{3}_{ml})_{\hat{\varphi}} (\tilde{A}^{1}_{ml'})_{\hat{\varphi}}]$$

$$R^{m}_{1l,3l'} = +4\pi \int_{0}^{+1} d\eta \Delta_n (\operatorname{Re} \tilde{t}^{1}_{ml})_{\xi} (\tilde{A}^{3}_{ml'})_{\xi}$$

To convert the above integrals into matrices one adopts the following convention for the index of the pair τl of vector and order indices in the case m=0, where negative indices are ignored:

$$n(\tau = 1, l) = 3(l - m) - 2$$

$$n(\tau = 2, l) = 3(l - m) - 1$$

$$n(\tau = 3, l) = 3(l - m)$$

In the case $m \neq 0$, the following index notation for the pair τl of vector and order indices is adopted:

 $n(\tau = 1, l) = 3(l - m) + 2$ $n(\tau = 2, l) = 3(l - m) + 1$ $n(\tau = 3, l) = 3(l - m)$

Here one uses the C-style index notation, where the first element in a vector starts at the index 0. The above index convention is used to project out the zero components of the vector basis functions in the case l=m=0. For example, the square R-matrix has dimension $(3(l_{max} - m) + 1)$ in the case m=0, and $(3(l_{max} - m) + 3)$ otherwise, where l_{max} is the maximum value for the order *l*.

The T-matrix is defined by the following equation in terms of the above matrices:

$$T = -(\operatorname{Re} QR^{-1}P - \operatorname{Re} M)(QR^{-1}P - M)^{-1}$$

APPENDIX J

COMPUTATION OF T-MATRIX FOR FINITE CYLINDER

Approved for public release; distribution is unlimited

t

J-1/J-2

APPENDIX J - COMPUTATION OF T-MATRIX FOR FINITE CYLINDER

This appendix describes the computation of the spheroidal T-matrix for a finite cylinder with hemi-spherical end caps.

First, the author begins with a discussion of the geometry of the surface of the cylinder. Let L denote the length of the cylinder and D its diameter, where the aspect ratio $\chi = (L/D)$ is greater than unity. The semi-focal distance of the spheroidal coordinate system is defined by the following equation in terms of the length and aspect ratio of the cylinder:

$$f = (L/2)\frac{\sqrt{\chi^2 - 1}}{\chi}$$

Define the quantity $L_{cyl} = L - D$ as the length of the cylindrical portion of the target excluding the end caps.

In the case $-L_{cyl}/2 < z < +L_{cyl}/2$, the normal to the surface is given by the following equation:

$$(n_{cyl})_{\xi} = +\xi \sqrt{\frac{1-\eta^2}{\xi^2 - \eta^2}}$$
$$(n_{cyl})_{\eta} = -\eta \sqrt{\frac{\xi^2 - 1}{\xi^2 - \eta^2}}$$

$$(n_{cvl})_{\hat{\omega}} = 0$$

The spheroidal coordinates of the point on the surface in the $\varphi = 0$ plane are given by the following equations:

 $\tilde{\rho} = \rho / f = D / f / 2$ $\tilde{z} = z / f$ $\tilde{r} = r / f = \sqrt{\tilde{\rho}^2 + \tilde{z}^2}$ $\xi = \sqrt{\frac{1 + \tilde{r}^2 + \sqrt{(1 + \tilde{r}^2)^2 - 4\tilde{z}^2}}{2}}$ $\eta = \tilde{z} / \xi$

To describe the surface of the target at its end caps one uses the polar angle $-\pi/2 < \vartheta < +\pi/2$ to parameterize the surface at its end caps. The coordinates of a point on the end caps in the $\varphi = 0$ plane are given by the following relationships:

$$\tilde{\rho} = (D/2f)\sin(\vartheta)$$

$$\tilde{z} = z/f = \pm ((D/2f)\cos(\vartheta) + L_{cyl}/2)$$

$$\tilde{r}^2 = \tilde{\rho}^2 + \tilde{z}^2$$

$$\xi = \sqrt{\frac{1 + \tilde{r}^2 + \sqrt{(1 + \tilde{r}^2)^2 - 4\tilde{z}^2}}{2}}$$

$$\eta = \tilde{z}/\xi$$

The normal to the surface at its end caps in the $\varphi = 0$ plane is given by the following equations:
$$x = f\sqrt{(\xi^{2} - 1)(1 - \eta^{2})}$$

$$z = f\xi\eta$$

$$n_{x} = \sin(\vartheta)$$

$$n_{z} = \cos(\vartheta)$$

$$n_{\xi} = n_{x}\partial_{\xi}x + n_{z}\partial_{\xi}z = f[+\xi\sqrt{\frac{1 - \eta^{2}}{\xi^{2} - 1}}\sin(\vartheta) + \eta\cos(\vartheta)]$$

$$n_{\eta} = n_{x}\partial_{\eta}x + n_{z}\partial_{\eta}z = f[-\eta\sqrt{\frac{\xi^{2} - 1}{1 - \eta^{2}}}\sin(\vartheta) + \xi\cos(\vartheta)]$$

$$n_{\varphi} = 0$$

$$n_{\xi} = [+\xi\sqrt{1 - \eta^{2}}\sin(\vartheta) + \eta\sqrt{\xi^{2} - 1}\cos(\vartheta)]/\sqrt{\xi^{2} - \eta^{2}}$$

$$n_{\xi} = \left[+\xi\sqrt{1-\eta^2}\sin(\vartheta) + \eta\sqrt{\xi^2 - 1}\cos(\vartheta) \right] / \sqrt{\xi^2 - \eta^2}$$
$$n_{\eta} = \left[-\eta\sqrt{\xi^2 - 1}\sin(\vartheta) + \xi\sqrt{1-\eta^2}\cos(\vartheta) \right] / \sqrt{\xi^2 - \eta^2}$$
$$n_{\phi} = 0$$

The component n_{ξ} of the normal has even parity under the transformation $\eta \rightarrow -\eta$, whereas the component $n_{\hat{\eta}}$ has odd parity. In the case of the normal for the end caps, the angle ϑ becomes $\vartheta \to \vartheta + \pi$ under the transformation $\eta \to -\eta$.

2

Since the finite cylinder is both symmetric under the transformation $\eta \rightarrow -\eta$ and $\varphi \rightarrow \varphi + \alpha$, the T-matrix is diagonal in parity σ and azimuthal order m. In addition, only terms for which the difference (l-l') is even have non-zero elements for the T-matrix. These symmetries allow us to reduce the two-dimensional surface integrals to a one-dimensional integral over the variable $0 > \eta > 1$. Due to the above symmetries, the T-matrix is of the following form:

 $T_{\sigma m l, \sigma' m' l'} = \delta_{\sigma}^{\sigma'} \delta_{m}^{m'} T_{l, l'}^{m}$

Define the following basis functions by replacing the trigonometric functions in the basis functions by the exponential functions $\exp[\pm im\varphi]$, respectively:

$$S^{W}_{\pm ml} \exp[\pm im\varphi] = \frac{1}{\sqrt{\varepsilon(m)}} (S^{W}_{eml} \pm iS^{W}_{oml})$$

$$A^{W}_{\pm ml} \exp[\pm im\varphi] = \frac{1}{\sqrt{\varepsilon(m)}} (A^{W}_{eml} \pm iA^{W}_{oml})$$

$$V^{W}_{\pm ml} \exp[\pm im\varphi] = \frac{1}{\sqrt{\varepsilon(m)}} (V^{W}_{eml} \pm iV^{W}_{oml})$$

$$A^{\tau}_{\pm ml} \exp[\pm im\varphi] = \frac{1}{\sqrt{\varepsilon(m)}} (A^{\tau}_{eml} \pm iA^{\tau}_{oml})$$

$$V^{\tau}_{\pm ml} \exp[\pm im\varphi] = \frac{1}{\sqrt{\varepsilon(m)}} (V^{\tau}_{eml} \pm iV^{\tau}_{oml})$$

$$t^{W}_{\pm ml} \exp[\pm im\varphi] = \frac{1}{\sqrt{\varepsilon(m)}} (t(V^{\psi}_{eml}) \pm it(V^{\psi}_{oml}))$$

$$t^{\tau}_{\pm ml} \exp[\pm im\varphi] = \frac{1}{\sqrt{\varepsilon(m)}} (t(V^{\psi}_{eml}) \pm it(V^{\psi}_{oml}))$$

The functions S^{W} , A^{W} , V^{W} , and t^{W} , are the scalar harmonic, vector harmonic, vector basis function, and traction vector in the external fluid, respectively. The functions A^{τ} , V^{τ} , and t^{τ} are the vector harmonics, vector basis functions, and traction vectors in the interior, respectively. Depending upon whether a given function $F_{\pm ml}$ contains an even or odd number of derivatives with respect to the angle φ , the function will be proportional to either "1" or $\pm i$.

Define the following notation:

 $F_{\pm ml} = \tilde{F}_{ml} \begin{cases} \pm 1, \text{ even azimuthal parity} \\ \pm i, \text{ odd azimuthal parity} \end{cases}$

Define the following functions based on their azimuthal parity:

$$\begin{split} & (\tilde{A}_{ml}^{1})_{\eta} = (A_{\pm ml}^{1})_{\eta} / \pm i \\ & (\tilde{A}_{ml}^{1})_{\varphi} = (A_{\pm ml}^{1})_{\varphi} \\ & (\tilde{A}_{ml}^{2})_{\eta} = (A_{\pm ml}^{2})_{\eta} \\ & (\tilde{A}_{ml}^{2})_{\varphi} = (A_{\pm ml}^{2})_{\varphi} / \pm i \\ & (\tilde{A}_{ml}^{3})_{\xi} = (A_{\pm ml}^{3})_{\xi} \end{split}$$

 $\tilde{S}_{ml}^{W} = S_{\pm ml}^{W}$

$$\begin{split} (\tilde{V}_{ml}^{1})_{\xi} &= (V_{\pm ml}^{1})_{\xi} / \pm i \\ (\tilde{V}_{ml}^{1})_{\eta} &= (V_{\pm ml}^{1})_{\eta} / \pm i \\ (\tilde{V}_{ml}^{1})_{\varphi} &= (V_{\pm ml}^{1})_{\varphi} \\ (\tilde{V}_{ml}^{2})_{\xi} &= (V_{\pm ml}^{2})_{\xi} \\ (\tilde{V}_{ml}^{2})_{\eta} &= (V_{\pm ml}^{2})_{\eta} \\ (\tilde{V}_{ml}^{2})_{\varphi} &= (V_{\pm ml}^{2})_{\varphi} / \pm i \\ (\tilde{V}_{ml}^{3})_{\xi} &= (V_{\pm ml}^{3})_{\xi} \\ (\tilde{V}_{ml}^{3})_{\eta} &= (V_{\pm ml}^{3})_{\eta} \\ (\tilde{V}_{ml}^{3})_{\varphi} &= (V_{\pm ml}^{3})_{\varphi} / \pm i \end{split}$$

۴

¥

ı

$$\begin{split} (\tilde{t}_{ml}^{1})_{\xi} &= (t_{\pm ml}^{1})_{\xi} / \pm i \\ (\tilde{t}_{ml}^{1})_{\eta} &= (t_{\pm ml}^{1})_{\eta} / \pm i \\ (\tilde{t}_{ml}^{1})_{\varphi} &= (t_{\pm ml}^{1})_{\varphi} \\ (\tilde{t}_{ml}^{2})_{\xi} &= (t_{\pm ml}^{2})_{\xi} \\ (\tilde{t}_{ml}^{2})_{\eta} &= (t_{\pm ml}^{2})_{\eta} \\ (\tilde{t}_{ml}^{2})_{\varphi} &= (t_{\pm ml}^{2})_{\varphi} / \pm i \\ (\tilde{t}_{ml}^{3})_{\xi} &= (t_{\pm ml}^{3})_{\xi} \\ (\tilde{t}_{ml}^{3})_{\eta} &= (t_{\pm ml}^{3})_{\eta} \\ (\tilde{t}_{ml}^{3})_{\varphi} &= (t_{\pm ml}^{3})_{\varphi} / \pm i \end{split}$$

The computation of the T-matrix requires the evaluation of the following surface integrals over the surface of the target:

$$Q_{n,\tau'n'} = -i\frac{k}{\lambda^{W}} \int_{S} dA\{(n \bullet t(V_{n}^{W})(n \bullet A_{\tau',n'}))\}$$
$$M_{n,n'} = -i\frac{k}{\lambda^{W}} \int_{S} dA\{(n \bullet V_{n}^{W})S^{W}_{n'}\}$$
$$P_{\tau n,n'} = \int_{S} dA\{(n \bullet \operatorname{Re} V_{\tau,n})S^{W}_{n'}\}$$
$$R_{\tau n,\tau'n'} = \int_{S} dA\{t(\operatorname{Re} V_{\tau n}) \bullet A_{\tau'n'}\}$$

In the above integrals one adopts the convention that the left hand term in the integral has +m and the right hand term has -m for the azimuthal angle dependence of the integrand. The above integrals are diagonal in m, and they can be replaced by the following one-dimensional integrals:

$$\begin{split} h_{\xi} &= f \sqrt{(\xi^{2} - \eta^{2})/(\xi^{2} - 1)} \\ h_{\eta} &= f \sqrt{(\xi^{2} - \eta^{2})/(1 - \eta^{2})} \\ h_{\varphi} &= f \sqrt{(\xi^{2} - 1)(1 - \eta^{2})} \\ \Delta_{n} &= h_{\eta}h_{\varphi}n_{\xi} - h_{\xi}h_{\varphi}n_{\eta} \frac{\partial\xi(\eta)}{\partial\eta} \\ Q^{m}_{l,\tau'l'} &= -2\pi i \frac{k}{\lambda^{W}} \int_{-1}^{+1} d\eta \Delta_{n} [(t^{W}_{+ml})_{\xi}n_{\xi} + (t^{W}_{+ml})_{\eta}n_{\eta}] [(A^{\tau'}_{-ml'})_{\xi}n_{\xi} + (A^{\tau'}_{-ml'})_{\eta}n_{\eta}] \\ M^{m}_{l,l'} &= -2\pi i \frac{k}{\lambda^{W}} \int_{-1}^{+1} d\eta \Delta_{n} [(V^{W}_{+ml})_{\xi}n_{\xi} + (V^{W}_{+ml})_{\eta}n_{\eta}] (S^{W}_{-ml'}) \\ P^{m}_{\tau l,l'} &= 2\pi \int_{-1}^{+1} d\eta \Delta_{n} [(\operatorname{Re} V^{\tau}_{+ml})_{\xi}n_{\xi} + (\operatorname{Re} V^{\tau}_{+ml})_{\eta}n_{\eta}] (S^{W}_{-ml'}) \\ R^{m}_{\tau l,\tau'l'} &= 2\pi \int_{-1}^{+1} d\eta \Delta_{n} (\operatorname{Re} t^{\tau}_{+ml}) \bullet (A^{\tau'}_{-ml'}) \end{split}$$

Using the polar parity and the azimuthal parity properties of the integrands, the non-zero components of the above integrals in the case l-l' is even are given by the following integrals:

$$R^{m}_{3l,3l'} = 4\pi \int_{0}^{+1} d\eta \Delta_n (\operatorname{Re} \tilde{t}^{3}_{ml})_{\xi} (\tilde{A}^{3}_{ml'})_{\xi}$$

$$R^{m}_{2l,3l'} = 4\pi \int_{0}^{+1} d\eta \Delta_n (\operatorname{Re} t^2_{+ml})_{\xi} (A^3_{+ml'})_{\xi}$$

•

ş.

ç

$$R^{m}_{3l,2l'} = 4\pi \int_{0}^{+1} d\eta \Delta_{n} [+ (\operatorname{Re}\tilde{t}^{3}_{ml})_{\hat{\eta}} (\tilde{A}^{2}_{ml'})_{\hat{\eta}} + (\operatorname{Re}\tilde{t}^{3}_{ml})_{\hat{\varphi}} (\tilde{A}^{2}_{ml'})_{\hat{\varphi}}]$$

J-9

The non-zero components of the integrals in the case l-l' is odd are given by the following integrals:

$$Q_{l,1l'}^{m} = -4\pi i (-i) \frac{k}{\lambda^{W}} \int_{0}^{+1} d\eta \Delta_{n} [+ (\tilde{t}_{ml}^{W})_{\xi} n_{\xi} + (\tilde{t}_{ml}^{W})_{\eta} n_{\eta}] [(\tilde{A}_{ml'}^{1})_{\eta} n_{\eta}]$$

$$P_{1l,l'}^{m} = +4\pi i \int_{0}^{+1} d\eta \Delta_{n} [+ (\operatorname{Re} \tilde{V}_{ml}^{1})_{\xi} n_{\xi} + (\operatorname{Re} \tilde{V}_{ml}^{1})_{\eta} n_{\eta}] (\tilde{S}_{ml'}^{W})$$

$$R_{1l,2l'}^{m} = +4\pi i \int_{0}^{+1} d\eta \Delta_{n} [+ (\operatorname{Re} \tilde{t}_{ml}^{1})_{\eta} (\tilde{A}_{ml'}^{2})_{\eta} - (\operatorname{Re} \tilde{t}_{ml}^{1})_{\phi} (\tilde{A}_{ml'}^{2})_{\phi}]$$

$$R^{m}_{2l,ll'} = -4\pi i \int_{0}^{+1} d\eta \Delta_{n} [+ (\operatorname{Re}\tilde{t}^{2}_{ml})_{\hat{\eta}} (\tilde{A}^{1}_{ml'})_{\hat{\eta}} - (\operatorname{Re}\tilde{t}^{2}_{ml})_{\hat{\varphi}} (\tilde{A}^{1}_{ml'})_{\hat{\varphi}}]$$

$$R^{m}_{3l,1l'} = -4\pi i \int_{0}^{+1} d\eta \Delta_{n} [+ (\operatorname{Re} \tilde{t}^{3}_{ml})_{\hat{\eta}} (\tilde{A}^{1}_{ml'})_{\hat{\eta}} - (\operatorname{Re} \tilde{t}^{3}_{ml})_{\hat{\varphi}} (\tilde{A}^{1}_{ml'})_{\hat{\varphi}}]$$

$$R^{m}_{1l,3l'} = +4\pi i \int_{0}^{+1} d\eta \Delta_n (\operatorname{Re} \tilde{t}^{1}_{ml})_{\xi} (\tilde{A}^{3}_{ml'})_{\xi}$$

In the case (l-l') is even, the components of the *R* and *P* matrices are real. In the case (l-l') is odd the *R* and *P* matrices are imaginary. The *R* and *P* matrices can be transformed into real matrices by making the following similarity transformation on the vector indices of the *R*, *P*, and *Q* matrices:

$$\begin{split} S_{\tau\tau'} &= -i\delta_{\tau}^{\tau'}\delta_{\tau}^{1} + \delta_{\tau}^{\tau'}\delta_{\tau}^{2} + \delta_{\tau}^{\tau'}\delta_{\tau}^{3} \\ S^{-i}_{\tau\tau'} &= S^{\dagger}_{\tau\tau'} = +i\delta_{\tau}^{\tau'}\delta_{\tau}^{1} + \delta_{\tau}^{\tau'}\delta_{\tau}^{2} + \delta_{\tau}^{\tau'}\delta_{\tau}^{3} \\ R^{m}_{\tau l,\tau' l'} &= \sum_{\tau^{*},\tau^{*}} S_{\tau',\tau^{*}}R^{m}_{\tau^{*}l,\tau^{m}l'}S^{\dagger}_{\tau^{*},\tau'} \\ P^{m}_{\tau l,l'} &= \sum_{\tau^{*}} S_{\tau',\tau^{*}}P^{m}_{\tau^{*}l,l'} \\ Q^{m}_{l,\tau' l'} &= \sum_{\tau^{*}} Q^{m}_{l,\tau^{*}l'}S^{\dagger}_{\tau^{*},\tau'} \end{split}$$

In the case (l-l') is even, the components of the Q, R, and P matrices are left invariant by this transformation, whereas in the case (l-l') is odd they are transformed to the following matrices:

$$Q^{m}_{l,ll'} = -4\pi i \frac{k}{\lambda^{W}} \int_{0}^{+1} d\eta \Delta_{n} [+(\tilde{t}^{W}_{ml})_{\hat{\xi}} n_{\hat{\xi}} + (\tilde{t}^{W}_{ml})_{\hat{\eta}} n_{\hat{\eta}}] [(\tilde{A}^{l}_{ml'})_{\hat{\eta}} n_{\hat{\eta}}]$$

$$P^{m}_{1l,l'} = +4\pi \int_{0}^{+1} d\eta \Delta_{n} [+(\operatorname{Re} \tilde{V}^{1}_{ml})_{\xi} n_{\xi} + (\operatorname{Re} \tilde{V}^{1}_{ml})_{\eta} n_{\eta}] (\tilde{S}^{W}_{ml'})$$

$$R^{m}_{1l,2l'} = +4\pi \int_{0}^{+1} d\eta \Delta_{n} [+ (\operatorname{Re}\tilde{t}^{1}_{ml})_{\hat{\eta}} (\tilde{A}^{2}_{ml'})_{\hat{\eta}} - (\operatorname{Re}\tilde{t}^{1}_{ml})_{\hat{\varphi}} (\tilde{A}^{2}_{ml'})_{\hat{\varphi}}]$$

$$R^{m}_{2l,1l'} = +4\pi \int_{0}^{+1} d\eta \Delta_{n} [+ (\operatorname{Re}\tilde{t}^{2}_{ml})_{\hat{\eta}} (\tilde{A}^{1}_{ml'})_{\hat{\eta}} - (\operatorname{Re}\tilde{t}^{2}_{ml})_{\hat{\varphi}} (\tilde{A}^{1}_{ml'})_{\hat{\varphi}}]$$

$$R^{m}_{3l,1l'} = +4\pi \int_{0}^{+1} d\eta \Delta_{n} [+ (\operatorname{Re} \tilde{t}^{3}_{ml})_{\hat{\eta}} (\tilde{A}^{1}_{ml'})_{\hat{\eta}} - (\operatorname{Re} \tilde{t}^{3}_{ml})_{\hat{\varphi}} (\tilde{A}^{1}_{ml'})_{\hat{\varphi}}]$$

$$R^{m}_{1l,3l'} = +4\pi \int_{0}^{+1} d\eta \Delta_{n} (\operatorname{Re} \tilde{t}^{1}_{ml})_{\xi} (\tilde{A}^{3}_{ml'})_{\xi}$$

To convert the above integrals into matrices one adopts the following convention for the index of the pair τl of vector and order indices in the case m=0, where negative indices are ignored:

$$n(\tau = 1, l) = 3(l - m) - 2$$

$$n(\tau = 2, l) = 3(l - m) - 1$$

$$n(\tau = 3, l) = 3(l - m)$$

L

4

٤.

In the case $m \neq 0$, one adopts the following index notation for the pair τl of vector and order indices.

$$n(\tau = 1, l) = 3(l - m) + 2$$

$$n(\tau = 2, l) = 3(l - m) + 1$$

$$n(\tau = 3, l) = 3(l - m)$$

Here one uses the C-style index notation, where the first element in a vector starts at the index 0. The above index convention is used to project out the zero components of the vector basis

functions in the case l=m=0. For example, the square R-matrix has dimension $(3(l_{max} - m) + 1)$ in the case m=0, and $(3(l_{max} - m) + 3)$ otherwise, where l_{max} is the maximum value for the order l.

The T-matrix is defined by the following equation in terms of the above matrices. $T = -(\operatorname{Re} QR^{-1}P - \operatorname{Re} M)(QR^{-1}P - M)^{-1}$

DISTRIBUTION

<u>COPIES</u>

1

₽

**

\$

<u>COPIES</u>

		ATTN: CODE 725 (RICHARD HOLTZAPPLE)	1
DOD ACTIVITIES (CONUS)		COMMANDING OFFICER	
		ATLANTIC FLEET WEAPONS TRAINING	
ATTN: CODE 32 (BRUCE JOHNSON)	1	FACILITY	
321 (JAN LINDBERG)	1	PSC 1008 BOX 3023	
3220M (TOM CURTIN)	1	FPO AA 34051-9000	
(JIM VALENTINE)	1		
32OE (TOM SWEAN)	1		
322W (DOUG TODOROFF)	1	NON-DOD ACTIVITIES (CONUS)	
OFFICE OF NAVAL RESEARCH			
800 NORTH QUINCY STREET		ATTN: KEVIN WILLIAMS	1
ARLINGTON VA 22217		APPLIED PHYSICS LAB	
		UNIVERSITY OF WASHINGTON	
ATTN: CODE 5582 (WILLIAM SMITH)	1	1013 NE 40 ST	
NAVAL RESEARCH LABORATORY		HN-10	
4555 OVERLOOK AVE SW		SEATTLE WA 98105	
WASHINGTON DC, 20375-5350			
		ATTN: LARRY GREEN	1
ATTN: JOHN HUGHES	1	APPLIED PHYSICS LABORATORY	
DEFENSE MODELING AND SIMULATION		JOHN HOPKINS UNIVERSITY	
OFFICE		JOHN HOPKINS RD	
1901 N BEAUREGARD ST		MAIL STOP 8-68	
STE 500		LAUREL MD 20723-6099	
ALEXANDRIA VA 22311			
		ATTN: RICHARD NIELSEN	1
ATTN: CODE 111 (JOHN SANCHIS)	1	BOEING COMPANY	
821 (ROBERT BENSON, JR.)	1	3370 MIRALOMA AVE	
8212 (MIKE MEDEIROS)	1	PO BOX 4921	
(JOHN IMPAGLIAZZO)	1	MC DC-27	
8221 (BOB CARPENTER)	1	ANAHEIM CA 92803-4921	
NAVAL UNDERSEA WARFARE CENTÉR			
1176 HOWELL ST		ATTN: ENSON CHANG	1
NEWPORT RI 02841-1708		DAVID MARX	1
		DYNAMIC TECHNOLOGY INC	
ATTN: CODE ONI-34 (BILL FEULNER)	1	21311 HAWTHRONE BLVD	
COMMANDER	•	STE 300	
OFFICE OF NAVAL INTELLIGENCE		TORRANCE CA 90503	
4521 SUITLAND RD			
WASHINGTON DC 20395-5720		ATTN: JOSE RIO	1
		LOCKHEED MARTIN FEDERAL SYSTEMS1	•
		9500 GODWIN DR	
		BLDG 250/060	
		MANASSAS VA 20110-4157	
		WIANASSAS VA 20110-413/	

DISTRIBUTION, CONTINUED

ų

A,

	<u>COPIES</u>		<u>COPIES</u>
ATTN: MEL PARRISH LOCKHEED MARTIN 199 BORTON LANDING DRIVE PO BOX 1027 MOORESTOWN NJ 08057	1	DEFENSE TECHNICAL INFORMATION CENTER 8725 JOHN J KINGMAN ROAD SUITE 0944 FT BELVOIR VA 22060-6218	2
ATTN: DONALD J BRIDY LOCKHEED MARTIN ELECTRONICS PARK BLDG 5 RM 300 MD 26 LIVERPOOL NY 13088	1	INTERNAL XPAOY (TECHNICAL LIBRARY) R20 (LATHROP) R21 (CHRISTOFE)	9 1
ATTN: CHARLES HAAS NORTHROP GRUMMAN ELECTRONIC SENSORS AND SYSTEMS SECTOR PO BOX 1488 ANNAPOLIS MD 21404	1	(COMMANDER) (FERNANDEZ) (LOPES) (MATTHEWS) (SAMMELMANN) R24 (MANNING)	1 1 1 3 1
ATTN: HERB FOX NORTHROP GRUMMAN OCEANIC SYSTEMS 895 OCEANIC DR MS 9030 ANNAPOLIS MD 21401	1		
ATTN: DOUGLAS ARENT RAYTHEON 1847 WEST MAIN RD MAILSTOP 146 PORTSMOUTH RI 02871	1		
ATTN: C. M. CIANY RAYTHEON NAVAL & MARINE INTEG SYS DIV BLDG 1W 1847 WEST MAIN RD MAIL STOP 174 PORTSMOUTH RI 02871	1		
ATTN: CHESTER LOGGINS SONATECH 879 WARD SANTA BARBARA CA 93111	1		

CLASSIFICATI DECLASSIFY WHEN () YES () NO	ON <u>UNCLASSIF</u> NMATERIAL IS DET (X) NOT APPLI	TED ACHED CABLE	
DOCUMENT TRANSMITTAL/REQUEST SHEET	COASTAL SYS PANAMA CITY	TEMS STATION, DAH) 7, FLORIDA 32407-7001	LGREN DIVISION
FROM: Commanding Officer Coastal Systems Station	ssic 3905	serial A94/04	DATE 05 Feb 2003
TO: Distribution	 () FOR INFORM COMPLETIO () FOR INFORM () IF MATERIAL ACCEPTABL () INTERLIBRA () RETURN OF (X) OTHER (SEI 	IATION AND DESTRUCTION N OF CITED CONTRACT/E IATION AND RETENTION L IS NOT AVAILABLE FOR E ON INTER-LIBRARY LO RY LOAN LOAN E BELOW)	ON UPON DELIVERY ORDER
SUBJECT: FORWARDING OF CSS/TR-02/18	<u> </u>		
REFERENCE:		а <u>_</u> , <u>, , , , , , , , , , , , , , , , , , </u>	
ENCL: (1) ACOUSTIC SCATTE TARGETS 1. Enclosure (1) is forwarded for d 2. The Coastal Systems Station point DSN 436-4442 or commercial (850) 23	RING FROM istribution t of contac 4-4442.	LARGE ASPECT : n. ct is Ann Alle	RATIO ELASTIC en (E53),
			- · ·
AUTHORIZED SIGNATURE G. W. Burch, By direction NCSC 5216/12 (REV 09-84)	1		
CLASSIFICATIO	DN <u>UNCLASSIF</u>	TIED	

÷.

DISTRIBUTION

COPIES

•

, t

, 2

COPIES

DOD ACTIVITIES (CONUS)		ATTN: CODE 725 (RICHARD HOLTZAPPLE) COMMANDING OFFICER	1
		ATLANTIC FLEET WEAPONS TRAINING	
ATTN: CODE 32 (BRUCE JOHNSON)	1	FACILITY	
321 (JAN LINDBERG)	1	PSC 1008 BOX 3023	
3220M (TOM CURTIN)	1	FPO AA 34051-9000	
(JIM VALENTINE)	1		
32OE (TOM SWEAN)	1		
322W (DOUG TODOROFF)	1	NON-DOD ACTIVITIES (CONUS)	
OFFICE OF NAVAL RESEARCH			
800 NORTH QUINCY STREET		ATTN: KEVIN WILLIAMS	1
ARLINGTON VA 22217		APPLIED PHYSICS LAB	_
		UNIVERSITY OF WASHINGTON	
ATTN: CODE 5582 (WILLIAM SMITH)	1	1013 NE 40 ST	
NAVAL RESEARCH LABORATORY		HN-10	
4555 OVERLOOK AVE SW		SEATTLE WA 98105	
WASHINGTON DC, 20375-5350			
		ATTN: LARRY GREEN	1
ATTN: JOHN HUGHES	1	APPLIED PHYSICS LABORATORY	-
DEFENSE MODELING AND SIMULATION		JOHN HOPKINS UNIVERSITY	
OFFICE		JOHN HOPKINS RD	
1901 N BEAUREGARD ST		MAIL STOP 8-68	
STE 500		LAUREL MD 20723-6099	
ALEXANDRIA VA 22311			
		ATTN: RICHARD NIELSEN	1
ATTN: CODE 111 (JOHN SANCHIS)	1	BOEING COMPANY	-
821 (ROBERT BENSON, JR.)	1	3370 MIRALOMA AVE	
8212 (MIKE MEDEIROS)	1	PO BOX 4921	
(JOHN IMPAGLIAZZO)	1	MC DC-27	
8221 (BOB CARPENTER)	1	ANAHEIM CA 92803-4921	
NAVAL UNDERSEA WARFARE CENTER			
1176 HOWELL ST		ATTN: ENSON CHANG	1
NEWPORT RI 02841-1708		DAVID MARX	ī
		DYNAMIC TECHNOLOGY INC	-
ATTN: CODE ONI-34 (BILL FEULNER)	1	21311 HAWTHRONE BLVD	
COMMANDER		STE 300	
OFFICE OF NAVAL INTELLIGENCE		TORRANCE CA 90503	
4521 SUITLAND RD			
WASHINGTON DC 20395-5720		ATTN: JOSE RIO	1
		LOCKHEED MARTIN FEDERAL SYSTEMS	-
		9500 GODWIN DR	
		BLDG 250/060	
		MANASSAS VA 20110-4157	

;*

DISTRIBUTION, CONTINUED

	<u>COPIES</u>		<u>COPIES</u>
ATTN: MEL PARRISH LOCKHEED MARTIN 199 BORTON LANDING DRIVE PO BOX 1027 MOORESTOWN NJ 08057	1	DEFENSE TECHNICAL INFORMATION CENTER 8725 JOHN J KINGMAN ROAD SUITE 0944 FT BELVOIR VA 22060-6218	2
ATTN: DONALD J BRIDY LOCKHEED MARTIN ELECTRONICS PARK	1	INTERNAL	
BLDG 5 RM 300 MD 26 LIVERPOOL NY 13088		XPAOY (TECHNICAL LIBRARY) R20 (LATHROP) R21 (CHRISTOFE)	9 1 1
ATTN: CHARLES HAAS NORTHROP GRUMMAN ELECTRONIC SENSORS AND SYSTEMS SECTOR PO BOX 1488 ANNAPOLIS MD 21404	1	R24 (CHRISTOFF) (COMMANDER) (FERNANDEZ) (LOPES) (MATTHEWS) (SAMMELMANN) R24 (MANNING)	1 1 1 3 1
ATTN: HERB FOX NORTHROP GRUMMAN OCEANIC SYSTEMS 895 OCEANIC DR MS 9030 ANNAPOLIS MD 21401	1		
ATTN: DOUGLAS ARENT RAYTHEON 1847 WEST MAIN RD MAILSTOP 146 PORTSMOUTH RI 02871	1		
ATTN: C. M. CIANY RAYTHEON NAVAL & MARINE INTEG SYS DIV BLDG 1W 1847 WEST MAIN RD MAIL STOP 174 PORTSMOUTH RI 02871	1		
ATTN: CHESTER LOGGINS SONATECH 879 WARD SANTA BARBARA CA 93111	1		